WHEN RECORDED RETURN TO:

Kennecott Land Company 5295 South 300 West, Suite 475 Murray, Utah 84107 Attention: Senior Contracts and Revenue Analyst 10179918 07/31/2007 04:24 PM \$202.00 Book - 9498 Pa - 1340-1421 GAF: Y W. OTT RECORDER, SALT LAKE COUNTY, UTAH FIRST AMERICAN TITLE BY: ZJM, DEPUTY - WI 82 P.

DECLARATION OF CONDOMINIUM

OF

EASTLAKE VILLAGE CONDOMINIUMS

IN

SALT LAKE COUNTY, UTAH

JULY 31 ,2007

Table of Contents

		Page
ARTIC	LE 1 DEFINITIONS	2
ARTIC	LE 2 CREATION OF THE CONDOMINIUM PROJECT	7
2.1	Submission	7
2.2	Name and Location	
2.3	Interpretation of Declaration and Applicability of the Act.	
2.4	Agent for Service of Process.	
ARTIC	LE 3 DESCRIPTION OF THE UNITS, LIMITED COMMON AREA COMMON AREA, ALLOCATED INTERESTS AND PLAT	8
3.1	Description of Boundaries of Each Unit and Unit Number	
3.2	Description of Limited Common Area for Parking	8
3.3	Description of Limited Common Area for Patios, Backyard, Balconies,	0
3.4	Entryways, Fireplaces and Storage Areas	
3.4	Division into Units, Minimum and Maximum Ownership Interests	
3.5	Allocated Interest of Each Unit in the Votes of the Association	
3.0	Allocated Interest of Each Unit in the Common Expenses of the Condominium Project	
3.7	Plat	
ARTICI	LE 4 MAINTENANCE AND UTILITIES	9
4.1	Maintenance of Units and Exclusive Limited Common Area	Q
4.2	Maintenance of Common Area and Nonexclusive Limited Common Area	
4.3	Default in Maintenance	
4.4	<u>Utilities</u>	
ARTICI	LE 5 MANAGEMENT	11
5.1	Organization of Association	11
5.2	Membership	
5.3	Voting	
5.4	Management Committee	
5.5	Qualification of Committee Members	
5.6	Action by Owners	
5.7	Annual Meeting	
5.8	Right of Association to Enter Units	
5.9	Association Rules.	
5.10		
5.11		

5.12	Availability of Condominium Project Documents	13
5.13	Managing Agent	14
5.14	Hearing before Management Committee	14
5.15	Declarant Control of Association.	14
ARTICLE	E 6 COVENANT FOR ASSESSMENTS	15
6.1	Creation of Lien and Personal Obligation for Assessment	15
6.2	Purpose of Assessments	
6.3	Regular Assessment	
6.4	Capital Improvement Assessments	
6.5	• • • • • • • • • • • • • • • • • • • •	
6.6	Percentage Assessments.	
6.7	Rules Regarding Billing and Collection Procedures	
	Certificate of Payment	
6.8	Special Assessments	
6.9	Date of Commencement of Assessments	
6.10	Application of Excess Assessments	
6.11	No Offsets	18
ARTICLE	E 7 EFFECT OF NONPAYMENT OF ASSESSMENTS AND	
ARTICLI	REMEDIES	10
		10
7.1	Due Date and Delinquency	18
7.2	Collection Charge	
7.3	Interest	
7.4	Action at Law	
7.5	Foreclosure Sale	
7.6	Suspension of Votes	
7.0	<u>Suspension of Yotes</u>	17
ARTICLE	E 8 PROPERTY RIGHTS IN COMMON AREA	19
0.1	Committee and the Committee an	1.0
8.1	General Easements to Common Area and Units.	
8.2	Public Utilities	
8.3	Easements for Encroachments	
8.4	Development Easements	
8.5	Limitation on Easement	
8.6	Party Walls	
8.7	Form for Conveyancing	
8.8	<u>Transfer of Title</u>	
8.9	<u>Views</u>	22
ARTICLE	E 9 USE RESTRICTIONS	22
9.1	Pulas and Pagulations	22
9.1	Rules and Regulations	
	Signs.	
9.3	Nuisance	
9.4	Temporary Structures	
9.5	Parking and Use of Open Parking/Visitor Parking	24

	9.6	External Fixtures	
	9.7	Window Covers	. 25
	9.8	External Laundering.	. 25
	9.9	Outside Speakers and Amplifiers	. 25
	9.10	Repairs	. 25
	9.11	Unsightly Items	. 25
	9.12	Oil and Mineral Activity	. 25
	9.13	Animals	. 25
	9.14	<u>Leases</u>	. 26
	9.15	Landscape Maintenance	. 26
	9.16	Floor Load	. 27
	9.17	Single Family Occupancy	. 27
	9.18	No Subdivision of Units or Further Restrictions	. 27
	9.19	Architectural Control	. 27
	9.20	<u>Lighting</u>	. 28
	9.21	Association Rules	. 28
	9.22	<u>Variances</u>	. 28
	9.23	<u>Hazardous Substances</u>	. 28
AR	TICLE	10 INSURANCE	. 29
	10.1		•
	10.1	Property Insurance	
	10.2	Comprehensive Public Liability Insurance	
	10.3	Workmen's Compensation Insurance	
	10.4	Fidelity Insurance	
	10.5	Premiums	
	10.6	Policy Provisions	
	10.7	Supplemental Insurance	
	10.8	Annual Insurance Report	
	10.9	Insurance Obtained by Owners	. 33
A D	TICI E	11 DESTRUCTION OF IMPROVEMENTS	24
AIN	TICLE	11 DESTRUCTION OF INTROVENES	. 34
	11.1	Reconstruction	34
	11.2	Reconstruction by Vote	
	11.3	Procedure for Minor Reconstruction	
	11.4	Procedure for Major Reconstruction	
	11.5	Determination not to Reconstruct Without Termination	
	11.6	Negotiations with Insurer	
	11.7	Repair of Units	
	11.8	Priority	
		/	
AR'	TICLE	12 EMINENT DOMAIN	. 36
	12.1	Total Taking of a Unit	36
	12.2	Partial Taking of a Unit.	
	12.2	Taking of Limited Common Area	37

12.4	Taking of Common Area	
12.5	Taking of Entire Condominium Project	37
12.6	Priority and Power of Attorney	37
ARTICLI	E 13 RIGHTS OF LENDERS	38
13.1	Notice of Lenders	38
13.2	Priority of Lenders	
13.3	Relationship with Assessment Liens	38
13.4	Required Lender Approval	39
13.5	Other Rights of Lenders.	40
13.6	Notices of Action	40
ARTICLE	E 14 TERMINATION	40
14.1	Required Vote	40
14.2	Termination Agreement	41
14.3	Sale of Condominium Project	
14.4	Association Duties	
14.5	Proceeds of Sale	41
ARTICLI	E 15 AMENDMENTS	41
15.1	Amendments by Declarant Prior to First Sale	
15.2	Amendments by Declarant After First Sale	
15.3	General Amendment Requirements	
15.4	Protection of Declarant Rights	
15.5	Execution of Amendments.	
15.6	Lender Approval	42
ARTICLI	E 16 EXPANSION	43
16.1	Option to Expand	43
16.2	Additional Land	43
16.3	Order of Exercise	43
16.4	<u>Limitations on Expansion</u>	43
16.5	Use Restrictions	
16.6	Nature of Improvements	
16.7	Substantial Completion	44
16.8	Documentation to Expand	44
16.9	Title to Units	
16.10	Conveyance of Additional Land	45
16.11	Amendment	45
16.12	Lender Approvals	15

ARTICLE 17 GENERAL PROVISIONS		45
17.1	Enforcement	45
17.2	No Waiver	
17.3	Cumulative Remedies	46
17.4	Severability	46
17.5	Covenants to Run with the Land: Term	
17.6	Allocation upon Termination	46
17.7	Construction	46
17.8	Gender and Number	47
17.9	Nuisance	47
17.10	Attorneys' Fees	47
17.11	Notices	
17.12	Effect of Declaration	
17.13	Personal Covenant	
17.14	Nonliability of Officials	48
17.15	Use of Funds Collected by the Association	48
17.16		
17.17	Owner Liability and Indemnification	
17.18	Conflicting Provisions	49
17.19	Consent, Power of Attorney, Waiver	
17.20	Security	50

EXHIBITS

- "A"
- Legal Description Parcel Legal Description Additional Land Form of Bylaws "B"
- "C"
- "D" Schedule of Units, Square Footage, Votes and Undivided Interests in Common Areas

DECLARATION OF CONDOMINIUM OF EASTLAKE VILLAGE CONDOMINIUMS (An Expandable Condominium Project)

THIS DECLARATION OF CONDOMINIUM ("Declaration") is made as of the date of the recording of this Declaration in the Salt Lake County Recorder's Office by KENNECOTT LAND COMPANY, a Delaware corporation ("**Declarant**"), pursuant to the Utah Condominium Ownership Act.

RECITALS

- A. On February 27, 2004, Declarant and Kennecott Land Residential Development Company, a Delaware corporation ("KLRDC") caused to be recorded that certain Community Charter for Daybreak, as Entry No. 8989518, in Book 8950, beginning at Page 7784, in the Official Records of Salt Lake County, Utah as amended by that certain Amendment No. 1 to Community Charter for Daybreak, recorded on August 26, 2004, as Entry No. 9156782, in Book 9030, beginning at Page 3767, in the Official Records of Salt Lake County, and as amended by that certain Amendment No. 2 to Community Charter for Daybreak, recorded on October 19, 2005, as Entry No. 9528104, in Book 9205, beginning at Page 4743, in the Official Records of Salt Lake County, and as amended by that certain Amendment No. 3 to Community Charter for Daybreak, recorded on March 13, 2007, as Entry No. 10031889, in Book 9434, beginning at Page 6476, in the Official Records of Salt Lake County (as may be amended from time to time, the "Master Residential Declaration"), to govern the phased development of a community commonly known as "Daybreak" located in the City of South Jordan, Utah, and further to govern the Daybreak Community Association, Inc. (the "Master Residential Association").
- B. In addition, on February 27, 2004, Declarant and KLRDC, caused to be recorded that certain Covenant for Community for Daybreak, as Entry No. 8989517, in Book 8950, beginning at page 7722, in the Official Records of Salt Lake County, Utah (as may be amended from time to time, the "Covenant"), to govern the phased development of Daybreak, and further to govern the Daybreak Community Council, Inc. (the "Daybreak Community Council").
- C. Pursuant to Section 2.6 of the Master Residential Declaration, Declarant may establish a separate owners' association to administer additional covenants applicable to a particular area within Daybreak.
- D. KLRDC is the fee owner of that certain real property (the "Parcel") situated in Salt Lake County, Utah, and located within Daybreak which is more particularly described in Exhibit "A," attached to and incorporated in this Declaration by this reference.
- E. Declarant and KLRDC desire to submit the Parcel, together with all buildings and improvements now or hereafter constructed on the Parcel, and all easements and rights appurtenant thereto (collectively, the "**Property**"), to the Utah Condominium Ownership Act as

a condominium project (the "Condominium Project") originally consisting of thirty (30) residential Units and related Common Area.

- F. Declarant and KLRDC desire to establish for their own benefit and for the mutual benefit of all future Owners or Occupants of the Condominium Project, certain covenants, conditions, restrictions, easements, rights, privileges, assessments and liens as set forth herein (collectively, the "**Restrictions**" which shall run with and be a burden upon the Property).
- G. Declarant and KLRDC intend that the Owners, Occupants, Lenders (as those terms are defined in Article 1 below) and all other persons hereafter acquiring any interest in the Property shall at all times enjoy the benefits of, and shall hold their interest subject to this Declaration, which is recorded in furtherance of establishing a general plan of condominium ownership for the Property, and for establishing rules for the use, occupancy, management and enjoyment thereof, all for the purpose of enhancing and protecting the value, desirability and attractiveness of the Condominium Project and the quality of life therein.
- H. Declarant reserves the right, as more fully set forth in Article 16 below, to expand the Condominium Project to include the Additional Land (as defined in Article 1) and the improvements thereon.
- I. The Condominium Project is and will be subject to the terms, conditions, covenants and restrictions set forth in the Master Residential Declaration and the Covenant, and will be governed by the Master Residential Association and the Daybreak Community Council.
- J. Holmes Homes, Inc., a Utah corporation ("**Builder**"), is purchasing the Property from KLRDC and plans to construct the Common Area and Units comprising the Condominium Project, all in accordance with the Plat.
- K. Daybreak Eastlake Village Condominiums Owners' Association, Inc. (the "Association"), is being created by filing Articles of Incorporation therefor ("Articles") with the Utah Division of Corporations and Commercial Code. The Association shall be the governing body of the Condominium Project created hereby and shall operate in accordance with the Association's Bylaws attached hereto as Exhibit "C" (the "Bylaws") and the Master Residential Declaration and the Covenant. The Association shall be a "Neighborhood Association" under the terms of the Master Residential Declaration.
- L. As more fully set forth in Article 16 hereof, Declarant reserves the right to expand the Condominium Project to include certain additional real property and improvements thereto.

NOW, THEREFORE, Declarant, and KLRDC, as owner of the Parcel, for the purposes above set forth, declare as follows:

ARTICLE 1 DEFINITIONS

1.1 Unless the context clearly indicates otherwise, certain terms used in this Declaration shall have the meanings set forth in this Article. All terms defined in the Recitals shall have the meanings set forth in the Recitals and in this Article.

- "Act" shall mean the Condominium Ownership Act, codified at Sections 57-8-1 through 57-8-38, Utah Code Annotated, pertaining to the creation, ownership and management of a condominium project in the State of Utah, as the same may be amended.
- **"Additional Builder"** shall mean a builder designated as such by the Declarant in a Supplemental Declaration pursuant to Section 16.10 of this Declaration.
- "Additional Land" shall mean all or any portion of the real property legally described on Exhibit "B," attached to and incorporated by reference in this Declaration.
 - "Allocated Interest" shall mean the interest in the Common Area appurtenant to a Unit.
- "Articles" shall mean the Articles of Incorporation by which the Association is formed under the Utah Revised Nonprofit Corporation Act, codified at Title 16, Chapter 6a, Utah Code Annotated, as the same may be amended.
- "Assessments" shall mean the charges against Owners to defray the Common Expenses as well as miscellaneous special assessments, special assessments for capital improvements, and special assessments for the purpose of restoring and reconstructing the Condominium Project in the event of casualty, all as provided in this Declaration.
- "Association" shall refer to Daybreak Eastlake Village Condominiums Owners' Association, Inc., whose membership shall include each Owner of a Unit in the Condominium Project, as required by the Act. The Association will be incorporated as a Utah nonprofit corporation prior to the conveyance of the first Unit in the Condominium Project by Builder or an Additional Builder.
- "Association Rules" shall mean and refer to the rules and regulations adopted by the Association pursuant to this Declaration and in furtherance of the Bylaws and Section 57-8-8 of the Act.
 - "Builder" shall mean Holmes Homes, Inc., a Utah corporation.
- **"Bylaws"** shall mean the Bylaws adopted by the Association pursuant to Section 57-8-15 of the Act for the purpose of regulating the affairs of the Association, as the same may be amended from time to time. The form of the Bylaws is attached to and incorporated by reference in this Declaration as Exhibit "C."
- **"Committee Member"** shall mean a duly qualified and elected or appointed member of the Management Committee.
- **"Common Area"** shall, unless otherwise provided in this Declaration or any Supplemental Declaration, mean:
 - (a) The land included within the Condominium Project, whether leasehold or in fee simple;

- (b) As applicable, the foundations, columns, girders, beams, supports, main walls, roofs, halls, corridors, lobbies, stairs, stairways, fire escapes, entrances, and exits of any buildings in the Condominium Project;
- (c) The basements, yards, gardens, parking areas, and storage spaces not part of the Limited Common Area or part of a Unit;
 - (d) The premises for lodging of janitors or persons in charge of the property;
- (e) Installations of central services such as power, light, gas, hot and cold water, heating, refrigeration, air conditioning, and incinerating;
- (f) As applicable, the elevators, tanks, pumps, motors, fans, compressors, ducts, and in general all apparatus and installations existing for common use;
- (g) All utility installations and all equipment connected with or in any way related to the furnishing of utilities to the Condominium Project and intended for the common use of all Owners, such as telephone, electricity, natural gas, water, cable television, and sewer:
- (h) A swimming pool, clubhouse, playground and such other community, recreational and commercial facilities, if any, as may be provided for in this Declaration; and
- (i) All other parts of the Condominium Project (excepting the Units) necessary or convenient to its existence, maintenance, and safety, or normally in common use.

The undivided interest in the Common Area appurtenant to each Unit is described in Section 3.4 of this Declaration and is set forth on Exhibit "D" attached to and incorporated in this Declaration by reference.

"Common Expenses" shall mean the actual and estimated costs for: (a) maintenance, management, operation, repair and replacement of the Common Area which is maintained by the Association; (b) deficiencies arising by reason of unpaid Assessments; (c) management and administration of the Association, including, but not limited to, compensation paid by the Association to managers, accountants, attorneys and employees; (d) utilities (other than separately metered utilities for the Units), including, common lighting and heating, water charges, trash collection and sewer service charges, extermination, security, gardening and other related services; (e) insurance and bonds required by this Declaration or any additional insurance and bonds obtained by the Management Committee in its discretion; (f) the establishment of reasonable reserves as the Management Committee shall deem appropriate in its discretion for the periodic maintenance, repair and replacement of the Common Area, which shall in no event be less than two (2) months of the estimated Assessments for each Unit; (g) expenses agreed upon as Common Expenses by the Association; and (h) other miscellaneous charges incurred by the Association or the Management Committee pursuant to the Act, this Declaration, the Bylaws or the Association Rules in furtherance of the purposes of the Association or in discharge of the duties and powers of the Association.

- "Community-Wide Standard" shall have the meaning assigned in the Master Residential Declaration.
- **"Condominium Project"** shall mean this real estate condominium project wherein fee simple title to single units in a multi-unit project, together with an undivided interest in the Common Area of the Property, are owned separately.
- **"Condominium Project DRC"** shall mean the design review committee for the Condominium Project, which is subject to the Design Review Committee described in the Master Residential Association.
- "Covenant" shall mean the Covenant for Community for Daybreak, recorded as Entry No. 8989517, in Book 8950, beginning at page 7722 in the Official Records of Salt Lake County, Utah.
- "Daybreak" shall mean the community commonly known as Daybreak located in South Jordan, Utah, subject to the Master Residential Declaration and/or the Covenant, as each may be expanded from time to time.
 - "Daybreak Community Council" shall mean the Daybreak Community Council, Inc.
- **"Declarant"** shall mean Kennecott Land Company, a Delaware corporation. Declarant is also the "Founder" referred to in the Master Residential Declaration and the Covenant.
- **"Declaration"** shall mean this Declaration, including all attached exhibits which are incorporated by reference, and any and all amendments and supplements to this Declaration.
- "Design Guidelines" shall have the meaning assigned in the Master Residential Declaration.
- **"Eligible Mortgagee"** shall mean and refer to a First Mortgagee which has requested notice of certain matters from the Association in accordance with Section 13.1 of this Declaration.
 - "FNMA" shall mean and refer to the Federal National Mortgage Association.
- **"First Mortgage"** shall mean any mortgage or deed of trust against a Unit which is not subject to any lien or encumbrance except liens for taxes or other liens which are given priority by statute.
- "First Mortgagee" shall mean any person named as a Lender under a First Mortgage, or any successor to the interest of any such person under a First Mortgage, which First Mortgage is not subject to any lien or encumbrance except liens for taxes or other liens which are given priority by statute.
- "Insurance Trustee" shall mean any trustee with which the Association may enter into an Insurance Trust Agreement, and which shall have exclusive authority to negotiate losses under the policies of insurance in accordance with such agreement.

5

- "KLRDC" shall mean Kennecott Land Residential Development Company, a Delaware corporation.
 - "Lender" shall mean a holder of a mortgage or deed of trust on a Unit.
- "Limited Common Area" shall mean a portion of the Common Area specifically designated as a Limited Common Area in this Declaration or the Plat and allocated by this Declaration or the Plat for the exclusive use of one or more Units to the exclusion of other Units.
- "Management Committee" shall mean the Management Committee of the Association elected pursuant to the Bylaws and serving as the management body of the Association.
- "Master Residential Association" shall mean the Daybreak Community Association, Inc.
 - "Master Residential Declaration" shall have the meaning assigned in Recital A hereof.
- "Occupant" shall mean a Person or Persons, other than an Owner, in possession of, or using a Unit, including, without limitation, family members, tenants, guests, or invitees.
- "Owner" shall mean the Person or Persons who are vested with record title of a Unit, and whose interest in the Unit is held in fee simple, according to the records of the County Recorder of Salt Lake County, Utah; however, Owner shall not include a Person who holds an interest in a Unit merely as security for the performance of an obligation. An Owner under this Declaration is also by definition an "Owner" under Section 2.4 of Master Residential Declaration and under Section 1.6 of the Covenant.
 - "Parcel" shall mean the real property legally described in Exhibit "A."
- **"Person"** shall mean a natural individual, corporation, estate, partnership, trustee, association, joint venture, government, governmental subdivision or agency or other legal entity capable of holding title to real property.
- "Plat" shall mean the plat constituting a record of survey map of the Property submitted with respect to the Condominium Project and entitled "Eastlake Village Condominiums Amending Parcels G, H, I, J, K, L, M, N, O & P as Shown on Kennecott Daybreak Plat 7 Subdivision", and recorded in the records of the County Recorder of Salt Lake County, Utah and all amendments thereto. "Plat" shall also refer to any additional plat which may be recorded with any Supplemental Declaration.
- "Property" shall mean the Parcel, together with all the buildings, improvements and permanent fixtures located thereon, and all easements and rights appurtenant thereto.
- "Restrictions" shall mean the covenants, conditions, assessments, easements, liens and restrictions set forth in this Declaration.

"Supplemental Declaration" shall mean a written instrument recorded in the records of the County Recorder of Salt Lake County, Utah, which refers to this Declaration and which amends, modifies, or supplements this Declaration in accordance with its terms.

"Unit" shall mean and refer to any of the separately numbered and individually described units now or hereafter shown on the Plat. Except where the context specifically requires otherwise, reference to a Unit shall include reference to the Allocated Interest in the Common Area appurtenant to such Unit. Each Unit shall be owned by the Owner or Owners of such Unit subject to all of the terms, covenants, conditions, restrictions and easements of this Declaration, the Master Residential Declaration and the Covenant. A Unit is included in the definition of a "Unit" under Section 3.1 of the Master Residential Declaration and Section 1.6 of the Covenant.

"Unit Number" shall mean the number, symbol or address that identifies one Unit in the Condominium Project.

ARTICLE 2 CREATION OF THE CONDOMINIUM PROJECT

- 2.1 <u>Submission</u>. Declarant, KLRDC, and Builder hereby submit and subject the Parcel to be a Condominium Project pursuant to the Act, and in furtherance thereof, makes and declares the Restrictions contained in this Declaration, and Declarant, KLRDC, and Builder hereby declare and agree that the Condominium Project and all of the Units shall be held, conveyed, transferred, sold, leased, mortgaged, encumbered, occupied, used and improved subject to the Restrictions, which Restrictions shall constitute covenants and conditions running with the land and shall be binding upon and inure to the benefit of Declarant, KLRDC, Builder, the Association, and each Owner, as their interests may appear, including their respective heirs, executors, administrators, personal representatives, successors and assigns. Each Owner acknowledges that the design and construction of the Units and all other initial improvements within the Project have been performed by Builder and not by Declarant or KLRDC and Owner has relied and will rely solely on Builder with respect to any issues related thereto.
- 2.2 Name and Location. The Condominium Project shall be named and known as EASTLAKE VILLAGE CONDOMINIUMS; provided, however, that in Declarant's sole discretion, the Condominium Project may be commonly or popularly referred to, for marketing, advertising, and other purposes, by a name or names other than Eastlake Village Condominiums. The Condominium Project is located in Salt Lake County, Utah, and the legal description of the real estate included in the Condominium Project is the Parcel set forth on Exhibit "A." The name of the Association is Eastlake Village Condominiums Owners' Association, Inc.
- 2.3 <u>Interpretation of Declaration and Applicability of the Act</u>. Declarant intends that the Condominium Project shall be governed by the Act, except where (in compliance with the Act) Declarant has included specific provisions in this Declaration which legally vary, supersede or supplement the Act, in which event such specific provisions of this Declaration which are contrary to the Act shall govern the Condominium Project.
- 2.4 Agent for Service of Process. Ty McCutcheon, located at 5295 South 300 West, Suite 475, Murray, Utah 84107 shall be the person to receive service of process for the

7

Condominium Project pursuant to Section 57-8-10(2)(d)(iii) of the Act, until such time as the Management Committee shall duly appoint a new agent and file a Supplemental Declaration.

ARTICLE 3 DESCRIPTION OF THE UNITS, LIMITED COMMON AREA COMMON AREA, ALLOCATED INTERESTS AND PLAT

- 3.1 <u>Description of Boundaries of Each Unit and Unit Number</u>. The cubic content space and Unit Number of each of the Units within the Condominium Project are set forth on the Plat. The vertical boundaries of each Unit shall be the underside of the finished but undecorated ceiling of the second floor of the Unit and the top of the finished but undecorated floor of the first floor or basement of the Unit, as applicable, as shown on the Plat. The horizontal boundaries of each Unit shall be the interior of the finished but undecorated walls located on the perimeter lines of the respective Units as shown on the Plat. All lath, furring, wallboard, plasterboard, plaster, paneling, tiles, wallpaper, paint, finished flooring and other materials constituting any part of the finished surfaces are part of the Unit, and all other materials constituting any part of the finished surfaces are part of the Unit, and all other and structural portions of the walls, floors, columns, or ceilings are part of the Common Area. If any pipes, wires, conduits, chutes, flues, ducts, shafts, public utility, water or sewer lines, or any other similar fixtures lies partially or wholly within or outside the designated boundaries of a Unit, any portion serving only that Unit is a Limited Common Area allocated solely to that Unit and any portion serving more than one Unit or any portion of the Common Area is part of the Common Area. Subject to the preceding sentence, each Unit shall consist of the space enclosed and bounded by the horizontal and vertical planes as shown on the Plat, and all interior partitions and other fixtures and improvements within such boundaries shall be part of the Unit. Conveyance of a Unit includes the use of the Limited Common Area appurtenant to the Unit.
- 3.2 <u>Description of Limited Common Area for Parking and Vehicular Access</u>. The parking space or spaces, if any, set forth on the Plat and designated for the respective Unit by corresponding number shall be an exclusive Limited Common Area for the Unit and such Owner. Such Limited Common Area shall be appurtenant to each respective Unit and may not be severed from the ownership of the Unit. The Limited Common Area depicted on the Plat that provides a Unit with vehicular access from a public right-of-way shall be appurtenant to all of the Units using such Limited Common Area for vehicular access from a public right-of-way.
- 3.3 <u>Description of Limited Common Area for Patios</u>, <u>Backyard</u>, <u>Balconies</u>, <u>Entryways</u>, <u>Fireplaces and Storage Areas</u>. The patio, porch, deck, backyard, balcony (or balconies), exterior screens and shutters and entryway, if any, which are adjacent to each respective Unit as set forth and depicted on the Plat shall be exclusive Limited Common Area for the Unit. The Limited Common Area shall be appurtenant to each respective Unit where so identified and may not be severed from the ownership of the Unit.
- 3.4 <u>Division into Units</u>, <u>Minimum and Maximum Ownership Interests</u>. The Condominium Project is hereby divided into 30 Units as set forth on the Plat, each such Unit consisting of a Unit and an equal appurtenant undivided interest in and to the Common Area. The Plat and/or Exhibit "D" hereto contain the Unit number and undivided interest of each Unit in the Condominium Project. If all of the Additional Land is added to the Condominium Project

pursuant to Article 16 of this Declaration, the maximum number of Units in the Project will be 600 Units and each Owner will have a minimum 1/600th undivided interest in the Common Area.

- 3.5 Allocated Interest of Each Unit in the Votes of the Association. Except as provided in Section 5.3 of this Declaration, the designation of the Allocated Interest which each Unit has in the votes of the Association is one vote for each Unit; thus, if the minimum number of Units in the Condominium Project is constructed, each Unit will have a 1/30th vote for all matters of the Association. The number of votes allocated to each Unit is set forth in Exhibit "D" attached hereto. In the event all of the Additional Land is added to the Condominium Project, each Unit will have a minimum 1/600th vote for all matters of the Association.
- 3.6 <u>Allocated Interest of Each Unit in the Common Expenses of the Condominium Project</u>. The designation of the Allocated Interest which each Unit bears in the Common Expenses of the Condominium Project is deemed to be equally divided; thus, if the minimum number of Units in the Condominium Project is constructed, each Unit will have a 1/30th Allocated Interest in the Condominium Project, each Unit will have a minimum 1/600th Allocated Interest in the Common Expenses.
- 3.7 <u>Plat</u>. The Plat is hereby incorporated into, and made an integral part of, this Declaration and all requirements and specifications set forth on the Plat and which are required by the Act are deemed included in this Declaration.

ARTICLE 4 MAINTENANCE AND UTILITIES

4.1 Maintenance of Units and Exclusive Limited Common Area. Each Owner shall furnish and be responsible for, at the Owner's own expense, all of the maintenance, repairs and replacements within the Owner's Unit and within any Limited Common Area appurtenant to the Owner's Unit, subject to Section 4.2(e) below. Such obligation shall include, without limitation: (a) the maintenance of all interior and exterior doors, including thresholds and door jams, all lath, furring, wallboard, plasterboard, plaster, paneling, tiles, wallpaper, paint, finished flooring, fireboxes of fireplaces and any other materials constituting the finished surfaces of floors, ceilings, or interior walls (all other portions of the walls, floors, or ceilings are part of the Common Area); (b) the repair and replacement of all windows, skylights, and door glass or equivalent materials and the interior and exterior cleaning of such windows and door glass; (c) the maintenance, in an open and unobstructed condition, of all sewer and drainage pipes, water and other utility lines serving an Owner's Unit between the points at which the same enter the Owner's Unit and the points where the same join the utility lines serving other Units; (d) the maintenance, replacement, repair and restoration of all of the following which serve an Owner's Unit exclusively: lighting fixtures (except exterior building mounted lights and walkway lights which are not located within patios and balconies), fans, plumbing fixtures, stoves, refrigerators, hot water heaters, air conditioning units (including compressors, condensers and forced air units), intercoms, security systems, and such other appliances, fixtures and decorations as an Owner may install; (e) the maintenance of the Unit and all exclusive Limited Common Area, such as enclosed parking, patios, backyards, fencing and balconies (including all materials above or upon the support structure, and railings and posts), exterior screens, shutters, and chimney

flues, that are within an Owner's exclusive control in a clean and sanitary condition, free of pests and rodents, and in good order and repair; and (f) the maintenance of the backyard within an Owner's exclusive control. An Owner may make nonstructural alterations within the Owner's Unit, but an Owner shall not make any structural or exterior alterations of the Common Area or the Limited Common Area without the prior written approval of the Management Committee.

Each Owner shall be liable to the Association or other Owners for damages to person or property in the Condominium Project caused by such Owner's negligence or the negligence of his family members, guests, visitors or invitees.

- 4.2 <u>Maintenance of Common Area and Nonexclusive Limited Common Area</u>. The Association, or its fully delegated representative, shall:
 - (a) Maintain and otherwise manage the Common Area and nonexclusive Limited Common Area, including, but not limited to, Condominium Project building exteriors, the landscaping, open areas and recreational facilities, if any, located thereon and maintain all open areas and exterior building mounted lights not within enclosed parking areas, patios and balconies, walkway and landscape area lights (located outside enclosed parking areas, patios and balconies), the structural support components of patios and roofs and flashings and other materials on patios and roofs that were installed to exclusively function as a roof, excluding skylights;
 - (b) Replace injured and diseased trees or other vegetation in any Common Area, and plant trees, shrubs and ground cover to the extent that the Management Committee deems necessary for the conservation of water and soil or for aesthetic purposes;
 - (c) Place and maintain upon any Common Area, such signs, markers and lights as the Management Committee may deem appropriate for the proper identification use and regulation thereof, subject to the approval of the Management Committee;
 - (d) Pay all electrical, water, natural gas, sanitary sewer, and other utility charges or fees for services furnished to the Common Area as the same become due and payable;
 - (e) Remove snow from sidewalks running throughout the Condominium Project and Limited Common Area driveways and walkways (whether exclusive or nonexclusive) immediately in front of each Unit; and
 - (f) Do all such other and further acts which the Management Committee deems necessary to preserve and protect the Common Area and the beauty thereof, in accordance with the general purposes specified in this Declaration.

The Association shall operate and maintain the sewer laterals associated with the Condominium Project. Billings received from the local sewer district will be paid by the Association. In addition, the Association shall be responsible for the repair and the damages associated with a sewer lateral back up as long as the cause originated in the lateral line and not in the trunk line.

The Management Committee shall determine, in its sole discretion, the appropriate maintenance of the Common Area. Provided, however, the Management Committee shall perform all

maintenance, repair and replacement obligations in a manner consistent with the Master Residential Declaration and the Community-Wide Standard. If the need for maintenance or repair is caused through the willful or negligent act of an Owner, the Owner's family, guests, licensees, lessees or invitees, the Management Committee may cause the maintenance or repair to be made. The Association shall invoice the Owner for the cost of such maintenance or repair. The Owner shall be deemed to have waived any objection to the invoice if the Owner does not, within thirty (30) days after receipt of the invoice, deliver a written objection to the Management Committee. The Owner's right to object is governed by the provisions of Section 5.14 of this Declaration. In the event the Management Committee determines that the Owner is liable for the maintenance or repair costs, the Association may enforce collection of such amounts as provided below for the collection of Assessments.

- 4.3 <u>Default in Maintenance</u>. If an Owner or Occupant fails to maintain a Unit or exclusive Limited Common Area pursuant to Section 4.1 above, or make repairs thereto in such a manner as may be deemed reasonably necessary in the judgment of the Management Committee to preserve and protect the attractive appearance and value of the Condominium Project, or if an Owner or Occupant fails to observe any Restrictions imposed on such Owner or Occupant by the terms of this Declaration, the Management Committee or its authorized representative shall give written notice to such Owner or Occupant stating with particularity the nature of the default and the corrective action which the Management Committee determines to be required and requesting that the same be carried out within a period of fourteen (14) days after the giving of such written notice. The Owner or Occupant shall be deemed to have waived any objection to the notice if the Owner or Occupant does not, within such fourteen (14) day period, deliver a written objection to the Management Committee. The Owner's or Occupant's right to object to the requested corrective action is governed by the provisions of Section 5.14 of this Declaration. If the Owner or Occupant fails to carry out such action within the period specified by the notice or as required following hearing before the Management Committee under Section 5.14, the Management Committee may cause corrective action to be taken and may levy a special Assessment for the cost thereof on the Owner. The special Assessment shall be due and payable within thirty (30) days after the Management Committee gives written notice thereof and shall be secured by the Assessment lien created in Section 6.1 of this Declaration.
- 4.4 <u>Utilities</u>. All utilities for individual Units (except those utility costs which are metered collectively and paid by the Association as a Common Expense item) will be metered separately to each Unit and such utility charges shall be the responsibility of the respective Owners.

ARTICLE 5 MANAGEMENT

5.1 Organization of Association. The Association will be organized no later than the date the first Unit in the Condominium Project is conveyed to an Owner other than Builder, any Additional Builder or Declarant. The Association shall serve as the governing body for all Owners. The Association shall make provision for the maintenance, repair, replacement, administration and operation of the Common Area, assessment of expenses, payment of losses, division of profits, acquisition of hazard insurance and disposition of such hazard insurance proceeds, and other matters as provided in the Act, this Declaration and the Bylaws. The

Association shall have all rights and powers granted to it under the Act and in this Declaration, the Articles and Bylaws. The Association shall not be deemed to be conducting a business of any kind, and all funds received by the Association shall be held and applied by it for the Owners in accordance with this Declaration, the Articles and the Bylaws.

- 5.2 Membership. Membership in the Association shall at all times consist exclusively of the Owners and each Owner shall be a member of the Association so long as such Owner owns a Unit and such membership shall automatically terminate when the Owner ceases to own a Unit. Upon the transfer of an ownership interest in a Unit, the new Owner succeeding to such ownership interest shall likewise succeed to such membership in the Association. If title to a Unit is held by more than one Person, the membership appurtenant to that Unit shall be shared by all such Persons in the same proportional interest and by the same type of tenancy in which title to the Unit is held. The Association shall make available to the Owners, Lenders and the holders, insurers and guarantors of the First Mortgage on any Unit current copies of this Declaration, Articles, Bylaws and other rules governing the Condominium Project and other books, records and financial statements of the Association. The term "available" as used in this Section 5.2 shall mean available for inspection, upon request, during normal business hours or under other reasonable circumstances.
- 5.3 <u>Voting</u>. At any meeting of the Association, each Owner of a Unit, including Declarant, either in person or by proxy, shall be entitled to one vote for each Unit owned. When more than one Person owns an interest in a Unit, such Persons shall designate to the Association, in writing, a representative who shall exercise the vote for such Unit on behalf of all co-Owners of the Unit. In no event shall fractional votes be exercised in respect to any Unit. The voting rights appurtenant to each Unit shall vest upon execution and recording of this Declaration and, except as otherwise permitted in this Declaration for expansion or otherwise, shall not be altered without the unanimous consent of all Owners expressed in a duly recorded Supplemental Declaration.
- 5.4 <u>Management Committee</u>. The governing body of the Association shall be the Management Committee elected pursuant to the Bylaws. Cumulative voting shall not apply for the purpose of electing Committee Members. The Management Committee shall consist of not less than three (3) members and not more than five (5) members. Except as otherwise provided in this Declaration, the Bylaws, or Association Rules, the Management Committee may act in all instances on behalf of the Association. The Management Committee may, as it deems appropriate, recommend amendments to the Bylaws and adopt, amend and repeal the Association Rules.
- 5.5 Qualification of Committee Members. Except for Committee Members elected or appointed by Declarant, each Committee Member shall be an Owner or the spouse of an Owner (or if an Owner is a corporation, partnership, limited liability company, or trust, a Committee Member may be an officer, partner, member, manager, trustee or beneficiary of such Owner). If a Committee Member ceases to meet such qualifications during the Committee Member's term, such person, in the discretion of the Management Committee, may continue to serve as a Committee Member until such Person's replacement has been duly qualified and approved by the remaining Committee Members. The replacement shall serve for the remainder of the departing Committee Member's term.

LA3:1127855.6 12

- 5.6 Action by Owners. Except as specifically provided herein, the Management Committee may not act on behalf of the Association to amend or terminate this Declaration, to elect Committee Members, except in filling vacancies in its membership for the unexpired portion of any term, or to determine the qualifications, powers and duties or terms of the Management Committee.
- 5.7 <u>Annual Meeting</u>. The Association shall conduct an annual meeting as provided in the Bylaws.
- 5.8 Right of Association to Enter Units. The Association acting through the Management Committee or its duly authorized agent shall have the right at all times upon reasonable notice (and at any time in case of an emergency) to enter upon or into any Unit to abate any infractions, to make repairs or correct any violation of any of the Restrictions herein set forth, and in connection therewith shall have the further right to assess all costs incurred against the Owner, such assessment to be secured by the lien provided in Section 6.1.
- 5.9 <u>Association Rules</u>. The Management Committee may adopt and administer the Association Rules in furtherance of the Bylaws for the regulation and operation of the Condominium Project.
- 5.10 Working Capital Fund. Declarant shall establish a working capital fund to meet unforeseen expenditures or to purchase any additional equipment or services needed for the operation of the Condominium Project. The initial amount of the working capital fund shall be at least equal to two (2) months of estimated Common Expenses for each Unit. Declarant shall collect these charges from each Owner at the time the sale of each Unit is closed. Any amounts collected and paid into this fund shall not be considered advance payments of Assessments. This fund cannot be used by Declarant to defray any of its expenses, reserve contributions or construction costs or to make up any budget deficits; provided, however, to the extent Declarant has paid the Association for an unsold Unit's share of this fund, Declarant shall be entitled to a reimbursement, to be paid at the time of disbursement out of the closing proceeds, for such amounts when such Unit is sold. Once control has been transferred to the Association, pursuant to Section 5.15, this fund shall be transferred to the Association.
- 5.11 Reserve Fund. The Association shall maintain an adequate reserve fund for maintenance, repair and replacement of the Common Area and nonexclusive Limited Common Area that must be replaced on a periodic basis, and such reserve shall be part of the monthly Assessments. To the extent the Management Committee deems necessary, surplus monies of the Association may be retained as additional reserves rather than refunded to the Owners or credited to future Assessments.
- 5.12 Availability of Condominium Project Documents. The Association will maintain current copies of this Declaration, the Articles, the Bylaws and the Association Rules concerning the Condominium Project and the Association's own books, records and financial statements available for inspection, upon request, during normal business hours by any Owner or Lender (or any insurer or guarantor of a Lender), and by any prospective purchaser of a Unit.

- 5.13 <u>Managing Agent</u>. The Management Committee may contract with a professional management agent to assist the Management Committee in the management and operation of the Condominium Project and may delegate such of its powers and duties to the management agent as it deems appropriate; provided, however, that only the Management Committee shall have the right to approve Association budgets, to impose a special Assessment and to authorize foreclosure of a lien for unpaid Assessments. Any management agreement must be terminable for cause upon thirty (30) days' notice, have a term not to exceed two years, and may be renewed with the consent of the manager and the Management Committee.
- 5.14 Hearing before Management Committee. In the event the Management Committee, under the terms of this Declaration, proposes to take action against an Owner, Occupant or Unit for a violation of the terms of this Declaration or the Association Rules, the Management Committee shall first give the applicable Owner or Occupant written notice of the proposed action and afford the Owner or Occupant the right to file an objection in writing to the proposed action. Unless otherwise provided in this Declaration, the Owner or Occupant shall have not less than twenty (20) days to respond in writing to the notice received from the Management Committee. If the Owner or Occupant timely objects to any proposed action by the Management Committee, the Management Committee shall, within the next twenty (20) days at a regular meeting of the Management Committee or at a special meeting convened for such purpose, consider the matter. The Owner or Occupant shall have an opportunity to appear before the Management Committee and provide testimony or evidence in support of the Owner's or Occupant's position. The Management Committee shall have authority to resolve the matter.
- 5.15 <u>Declarant Control of Association</u>. There is hereby established a period of Declarant control of the Association, during which period Declarant or persons designated by Declarant shall have the authority to appoint and remove the Association officers and Committee Members. The period of Declarant control shall terminate no later than the earlier of:
 - (a) The last to occur of (i) the date by which seventy-five percent (75%) of the Units have been conveyed to Unit purchasers; or (ii) all the Additional Land has been added to the Condominium Project; or
 - (b) Six (6) years after the first Unit is conveyed to an Owner; or
 - (c) When, in its discretion, the Declarant so determines and declares in a written instrument.

Notwithstanding anything in this Declaration that may be construed to the contrary with respect to Declarant's ability to voluntarily terminate its period of control of the Association as set forth above, Declarant may (in the exercise of its sole and absolute discretion) voluntarily terminate its period of control in whole or in part, with respect to all or any portion of any Unit, any Common Area, any Limited Common Area, any portion of the Additional Land, or with respect to any issue, matter or subject whatsoever. Declarant's decision to voluntarily terminate its period of control with respect to all or any portion of any Unit, Common Area, Limited Common Area, portion of the Additional Land, or with respect to any issue, matter or subject shall in no event affect, modify, or act to waive its authority under the period of its control of the

14

Association except with respect to such Unit, Common Area, Limited Common Area, portion of the Additional Land, or such issue, matter or subject.

ARTICLE 6 COVENANT FOR ASSESSMENTS

- 6.1 Creation of Lien and Personal Obligation for Assessment. Each Owner of any Unit, by acceptance of a deed or other instrument creating in such Owner the interest required to be deemed an Owner, whether or not it shall be so expressed in any such deed or other instrument, is deemed to covenant and agree with each other and with the Association to pay to the Association such Assessments to be fixed, established and collected from time to time as provided in this Declaration. The Assessments, together with interest thereon, collection charges, attorneys' fees, court costs and other costs of collection as hereinafter provided shall be secured by a continuing lien upon the Unit against which each such Assessment is made in favor of the Association. Each such Assessment, together with such interest, collection charges, costs and attorneys' fees, shall also be the personal obligation of the Owner of such Unit at the time the Assessment becomes due. However, a lien to secure unpaid Assessments shall not be affected by the sale or transfer of the Unit unless foreclosure by a prior recorded encumbrance is involved, in which case: (a) the foreclosure will extinguish the lien for any Assessments that were payable before the foreclosure sale, but shall not relieve any subsequent Owner from paying further Assessments; and (b) the prorated share of any extinguished Association lien may be redistributed to the other Units in the Condominium Project.
- 6.2 <u>Purpose of Assessments</u>. The Assessments levied by the Association shall be used exclusively for the purposes of promoting the health, safety and welfare of the Owners, the management, maintenance, care, preservation and protection of the Condominium Project, enhancing the quality of life in the Condominium Project and the value of the Condominium Project including, without limitation, the improvement and maintenance of the services and facilities devoted to this purpose and related to the use and enjoyment of the Common Area, or in furtherance of any other duty or power of the Association.
- 6.3 Regular Assessment. The Management Committee is expressly authorized to adopt and amend budgets from time to time. Not later than thirty (30) days prior to the beginning of each fiscal year, the Management Committee shall adopt a pro forma operating statement or budget for the upcoming fiscal year which shall, among other things, estimate the total Common Expenses to be incurred for such fiscal year. The Management Committee may, but is not required to, send a written summary of the budget to all Owners within thirty (30) days after the adoption of the proposed budget. The Management Committee shall at that time determine the amount of the regular Assessments to be paid by each Owner. Each Owner shall thereafter pay to the Association the Owner's regular Assessment in equal monthly installments on the first day of each month. In the event the Management Committee determines that the estimate of total charges for the current year is, or will become, inadequate to meet all Common Expenses for any reason, it shall then immediately determine the approximate amount of such inadequacy and issue a supplemental estimate of the Common Expenses and determine the revised amount of the regular Assessment against each Owner, and the date or dates when due. The Owners shall have no right to ratify any budget, or amendment thereof, adopted by the Management Committee.

- 6.4 <u>Capital Improvement Assessments</u>. In addition to regular Assessments, the Management Committee may levy in any fiscal year a capital improvement Assessment applicable to that year only for the purpose of defraying, in whole or in part, the cost of any construction or replacement (other than due to destruction) of a described capital improvement upon the Common Area, including the fixtures and personal property related thereto. The Management Committee shall not impose a capital improvement Assessment exceeding ten percent (10%) of the then estimated annual Common Expenses without the approval of Owners holding a majority of the Allocated Interests in the votes of the Association. All amounts collected as capital improvement Assessments may only be used for capital improvements and shall be deposited by the Management Committee in a separate bank account or the Association's account for reserve funds, to be held in trust for such purposes and said funds shall not be commingled with any other funds (other than reserve funds) of the Association.
- 6.5 <u>Percentage Assessments</u>. Except as otherwise provided herein, all Assessments (other than special Assessments) shall be in an amount based on the percentage interest for each Unit stated in Section 3.6 of this Declaration, as the same may be amended from time to time.
- 6.6 Rules Regarding Billing and Collection Procedures. The Management Committee shall have the right and responsibility to adopt rules and regulations setting forth procedures for the purpose of making the Assessments provided in this Declaration and for the billing and collection of regular and special Assessments, provided that such procedures are not inconsistent with the provisions hereof. The failure of the Association to send a bill to an Owner shall not relieve any Owner of liability for any Assessment or charge under this Declaration, but the Assessment lien therefor shall not be foreclosed or otherwise enforced until the Owner has been given not less than thirty (30) days' written notice prior to such foreclosure or enforcement, at the address of the Owner on the records of the Association, that the Assessment or any installment thereof is or will be due and of the amount owing. Such notice may be given at any time prior to or after delinquency of such payment.
- 6.7 <u>Certificate of Payment</u>. The Association shall, within ten (10) business days after written demand, furnish to any Owner liable for Assessments or to the Lender or a potential Lender for such Unit, a recordable written statement or certificate signed by an officer or authorized agent of the Association setting forth whether the Assessments relating to a specified Unit have been paid and the amount of delinquency, if any. To the extent permitted by law, a reasonable charge not to exceed twenty dollars (\$20) may be collected by the Management Committee for the issuance of each such certificate. Each certificate shall be conclusive evidence of payment of any Assessment therein stated to have been paid.
- 6.8 <u>Special Assessments</u>. Special Assessments shall be levied by the Management Committee against a Unit and its Owner to reimburse the Association for:
 - (a) Costs incurred in bringing an Owner or the Owner's Unit into compliance with the provisions of the Declaration, the Articles, the Bylaws or the Association Rules;
 - (b) Costs associated with the maintenance, repair or replacement of a Limited Common Area assigned to such Unit;

LA3:1127855,6

- (c) Any other charge designated as a special Assessment in this Declaration, the Articles, the Bylaws or the Association Rules; and
- (d) Attorneys' fees, interest and other charges relating thereto as provided in this Declaration.

In the event the Association undertakes to provide materials or services which benefit individual Units and which can be accepted or not by individual Owners, such Owners, in accepting such materials or services, agree that the costs thereof shall be a special Assessment.

6.9 Date of Commencement of Assessments.

- (a) <u>Condominium Project excluding Additional Land</u>. Regular and special Assessments as to Units within the Condominium Project (but not including Units within any Additional Land) for which construction has been substantially completed shall commence as to all such substantially completed Units on the first day of the month following the conveyance of the first Unit by Builder to an Owner. Thereafter, regular and other Assessments shall commence as to newly completed Units within the Condominium Project (but not including Units within any Additional Land) on the first day of the month following the substantial completion of construction for each respective Unit. Until the Association makes an Assessment, Builder shall pay all Common Expenses of the Association. No Assessments shall be payable on Units for which construction has not been substantially completed, provided, however, that Builder shall be obligated to pay to the Association any deficiencies in monies available to pay Common Expenses due to Builder not having paid an Assessment on uncompleted Units (except for Units within any portion of the Additional Land) and which are necessary for the Association to be able to pay all Common Expenses in a timely manner.
- (b) Condominium Project on Additional Land. Regular and special Assessments as to Units within any portion of the Additional Land for which construction has been substantially completed shall commence as to all such substantially completed Units in such portion of the Additional Land on the first day of the month following the conveyance of the first Unit by any Additional Builder for such portion of the Additional Land to an Owner. Thereafter, regular and other Assessments shall commence as to newly completed Units within such portion of the Additional Land on the first day of the month following the substantial completion of construction for each respective Unit. No Assessments shall be payable on Units within any portion of the Additional Land for which construction has not been substantially completed, provided, however, that the Additional Builder for such portion of the Additional Land shall be obligated to pay to the Association any deficiencies in monies available to pay Common Expenses due to such Additional Builder not having paid an Assessment on uncompleted Units within such portion of the Additional Land and which are necessary for the Association to be able to pay all Common Expenses in a timely manner.
- 6.10 <u>Application of Excess Assessments</u>. In the event the amount budgeted to meet Common Expenses for a particular fiscal year proves to be excessive in light of the actual Common Expenses, the Management Committee in its discretion may apply the excess to

reserves, credit the excess against future Assessments or pay the excess to the Owners in proportion to the Allocated Interests of each Unit in the Common Expenses of the Condominium Project, as the Management Committee deems appropriate. The decision of the Management Committee shall be binding and conclusive. In addition, the Association shall not be obligated to reduce the amount of Assessments in succeeding years if an excess exists for a prior year.

6.11 <u>No Offsets</u>. All Assessments shall be payable in the amount specified by the Assessment and no offsets against such amount shall be permitted for any reason, including, without limitation, a claim that the Association is not properly exercising its duties and powers as provided in this Declaration.

ARTICLE 7 EFFECT OF NONPAYMENT OF ASSESSMENTS AND REMEDIES

- 7.1 <u>Due Date and Delinquency</u>. The first day of each month shall be the Assessment due date for that month. Any Assessment which is not paid within thirty (30) days after it becomes due shall be delinquent. Whenever an Assessment is delinquent, the Management Committee may at its option invoke any or all of the sanctions granted in this Article 7.
- 7.2 <u>Collection Charge</u>. If any Assessment is delinquent, the Owner shall be obligated to pay the collection charge then provided for in the Bylaws or the Association Rules. The amount of such collection charge until paid shall constitute part of the Assessment lien as provided for in Section 6.1 of this Declaration.
- 7.3 <u>Interest</u>. If any Assessment is delinquent, interest at the rate of twelve percent (12%) per annum may be assessed on the amount owing from the date due until such time as it is paid.
- 7.4 Action at Law. The Association may bring an action to recover a delinquent Assessment either personally against the Owner obligated to pay the same or foreclose the Assessment lien; provided, however, the Owner shall have the right to object to the nature or amount of a delinquent Assessment as provided in Section 5.14 of this Declaration. In addition, the Association's choice of one remedy shall not prejudice or constitute a waiver of the Association's right to exercise any other remedy. The costs of preparing and filing the complaint shall be assessed against the delinquent Owner and the Owner's Unit and reasonable attorneys' fees and court costs will thereafter be added to the amount in delinquency (plus interest and collection charges, if appropriate) in the event that a judgment is obtained by the Association. Each Owner vests in the Association or its successors or assigns the right and power to bring actions at law or lien foreclosures against such Owner or Owners for the collection of delinquent Assessments.
- 7.5 Foreclosure Sale. Any foreclosure provided for in this Declaration is to be conducted in compliance with applicable provisions relating to the foreclosures of deeds of trust or realty mortgages in the State of Utah. In any foreclosure or sale, the Owner of the affected Unit shall pay the costs and expenses of such proceedings and reasonable attorneys' fees. The Association, upon approval by a majority of the Allocated Interests in the votes of the Association, may through its duly authorized agents have and exercise the power to bid on the Unit at the foreclosure or other sale and to acquire, hold, lease, mortgage and convey such Unit.

7.6 <u>Suspension of Votes</u>. The Management Committee may suspend the obligated Owner's right to vote on any matter at regular or special meetings of the Association and the Owner's right to use all or any portion of the Common Area (exclusive of the Limited Common Area appurtenant to the Owner's Unit) for the entire period during which an Assessment or other amount due under any of the provisions of the Declaration remains delinquent. Prior to suspending the Owner's right to vote, the Management Committee shall afford the Owner the right to a hearing before the Management Committee as provided in Section 5.14 of this Declaration.

ARTICLE 8 PROPERTY RIGHTS IN COMMON AREA

8.1 General Easements to Common Area and Units.

- (a) Subject to this Declaration and the Association Rules, each Owner shall have an equal undivided interest, right and easement of use and enjoyment in and to the Common Area. Each Owner shall have an unrestricted right of ingress or egress to and from the Owner's Unit over and across such Common Area (exclusive of the Limited Common Area), and the nonexclusive right to the use of all open parking stalls, if any, within the Common Area. Each Owner shall also have the exclusive right to use and enjoy any Limited Common Area that may be designated for exclusive use by such Owner. Such rights and easements shall be appurtenant to and shall pass with title to each Unit and in no event shall such appurtenant rights be separated therefrom. Any Owner may delegate the right and easement of use and enjoyment described herein to any family members, household guest, tenant, lessee, contract purchaser, Occupant or other Person who resides in such Owner's Unit.
- (b) The Association, acting through the Management Committee or its authorized agent, and public utility companies providing service to the Condominium Project, shall have nonexclusive easements with the right of access to each Unit to make inspections, to maintain, repair, replace or effectuate the restoration of the Common Area accessible from such Unit. Such rights shall be exercised in a reasonable manner and at reasonable times with prior notification, unless emergency situations demand immediate access. The Association, acting through the Management Committee or its authorized agent, shall have a nonexclusive right to grant permits, licenses and easements upon, across, over, under and through the Common Area for purposes necessary for the proper operation of the Condominium Project; provided, however, such rights shall be exercised in a reasonable manner and at reasonable times with proper notification, unless emergency situations demand immediate access.
- (c) Declarant hereby reserves unto itself easements and rights of way as set forth in Chapter 13 of the Master Residential Declaration. Further, the Condominium Project shall continue to be subject to the applicable easements set forth in the Master Residential Declaration.
- 8.2 <u>Public Utilities</u>. Easements and rights-of-way over the Condominium Project for the installation and maintenance of electricity lines, telephone lines, cable television, water lines, gas

lines, sanitary sewer lines, drainage facilities, and such other public utilities needed to serve the Condominium Project are hereby granted to the Association, together with the right to grant and transfer the same; provided, however, such easements and rights-of-way shall not unreasonably interfere with the use of the Common Area and the Units by the Owners or Occupants. The Association shall have the power to grant and convey, in the name of all of the Owners as their attorney-in-fact, to any other person easements and rights-of-way in, on, over or under the Common Area for the purpose of constructing, erecting, operating or maintaining lines, cables, wires, conduits, or other devices for electricity, cable television, power, telephone, public sewers, storm water drains and pipes, water systems, sprinkling systems, water heating and gas lines or pipes and any similar public or quasi public improvements or facilities, and each Owner in accepting the deed to a Unit expressly consents to such easements and rights-of-way and authorizes and appoints the Association as attorney-in-fact for such Owner to execute any and all instruments conveying or creating such easements or right-of-way. Such Owner and those claiming by, through or under an Owner agree to execute promptly all such documents and instruments and to do such other things as may be necessary or convenient to effect the same at the request of the Association. However, no such easement can be granted if it would permanently interfere with the use, occupancy or enjoyment by any Owner or such Owner's Unit.

- 8.3 Easements for Encroachments. If any portion of the Common Area now encroaches upon any Unit, or if any Unit now encroaches upon any other Unit or the Common Area, or if any such encroachment shall occur hereafter as a result of the manner in which the buildings are constructed or due to settling, shifting, alteration, replacement, repair or restoration by Declarant or the Association, a valid easement for encroachment, and maintenance of such encroachment, shall exist so long as the buildings stand.
- 8.4 <u>Development Easements</u>. Until all Units have been sold by Builder or any Additional Builder to third-party purchasers, there are hereby reserved to Declarant, Builder or any such Additional Builder or their respective assignee, together with the right to grant and transfer the same to others, including their respective sales agents, representatives, successors and assigns, easements and rights upon, across, over, under and through the Condominium Project:
 - (a) For the purpose of completing all improvements contemplated by this Declaration and the Plat, including, but not limited to, improvements to the Additional Land;
 - (b) To use any Unit owned by Builder, an Additional Builder or Declarant or any other Unit, with the express permission by the Owner of such Unit, as a sales model, or as a management, maintenance or sales office (including the use of signs and other advertising devices) in connection with the erection, remodeling and sale or lease of Units within the Condominium Project, and to relocate such models and offices to other Units or Common Areas at any time; and
 - (c) To create and cause noise, dust, vibration and other nuisances or annoyances created by and resulting from any work connected with or incidental to the development, construction and sale of any Unit or other improvement to the Condominium Project or Daybreak, and each Owner or Occupant in such Owner's Unit waives any and all rights,

claims or actions that might otherwise be asserted against Declarant, Builder or any Additional Builder, their agents, employees, licensees, successors and assigns, based on any such noise, dust, vibration and other nuisances or annoyances;

provided, however, that no such use by Declarant, Builder or any Additional Builder, or their agents shall otherwise restrict Owners or Occupants in the reasonable use and enjoyment of their Units.

- 8.5 <u>Limitation on Easement</u>. An Owner's undivided interest in, and right and easement of use and enjoyment concerning, the Common Area shall be subject to the following:
 - (a) The right of the Association, subject to the provisions of Section 5.14 of this Declaration, to suspend the Owner's voting right in the Association and the Owner's right to the use of any recreational facilities included in the Common Area for any period during which (i) an assessment on such Owner's Unit remains unpaid; (ii) for a period not exceeding sixty (60) days for any infraction by such Owner of the provisions of this Declaration or any Association Rule; and (iii) for successive 60-day periods if any such infraction is not corrected during any prior 60-day suspension period;
 - (b) The right of the Association to impose reasonable limitations on the number of guests per Owner or Occupant who at any given time are permitted to use the Common Area; and
 - (c) The right of any governmental or quasi-governmental body having jurisdiction over the Property to access and to have rights of ingress and egress over and across any street, parking areas, walkway or open areas contained within the Common Area for purposes of providing police and fire protection, transporting school children and providing other governmental or municipal services.
- 8.6 Party Walls. Each wall or ceiling-floor physical boundary which is built as part of the original construction (or reconstruction) of the Units and placed on the dividing line (whether horizontal or vertical) between the Units shall constitute a "Party Wall" and, to the extent not inconsistent with the provisions of this Section 8.6, the general rules of law regarding party walls and liability for property damage due to negligence or willful acts or omissions shall apply to all Party Walls within the Condominium Project. The cost of reasonable repair and maintenance of a Party Wall shall be shared equally by the Owners who make use of such Party Wall. If a Party Wall is destroyed or damaged by fire or other casualty, the provisions of Article 11 shall apply. Notwithstanding any other provision of this Section 8.6, an Owner who by negligent or willful act causes a Party Wall to be damaged shall, subject to the provisions of Section 5.14 of this Declaration, bear the entire cost of furnishing repairs to the Party Wall. The right of any Owner to contribution from any other Owner under this Section 8.6 shall be appurtenant to the land and shall pass to such Owner's successors in title.
- 8.7 Form for Conveyancing. Any deed, lease, mortgage, deed of trust or other instrument conveying or encumbering title to a Unit may describe the interest or estate involved substantially as follows:

Building No and Unit No Condominiums, as said Unit is identifi	of Eastlake Village
development recorded	2007, as Entry Number
in Book <u>9498</u> , at Page	1339 of the official records of
the Salt Lake County Recorder, State of	of Utah, and as identified and
described in the Declaration of Condor	
Condominiums, an Expandable Condo	
, 2007, as Entry Numbe	r, in Book
at Page of the official records of	f the Salt Lake County
Recorder, State of Utah. TOGETHER	
and a right and easement of use and en	joyment, to the Common Area
described in, as provided for, and in the	
Declaration This conveyance is subje	
Declaration, including any amendment	s thereto. The undivided
interest in the Common Area conveyed	l hereby is subject to
modification, from time to time, as pro	vided in the Declaration for
expansion of the Condominium Projec	t.

Whether or not the description employed in any such instrument is in the above specified form, however, all provisions of this Declaration shall be binding upon and shall inure to the benefit of any party who acquires any interest in a Unit.

- 8.8 <u>Transfer of Title</u>. At the time of the first conveyance of each Unit from Builder or an Additional Builder to a third-party purchaser, Builder or such Additional Builder shall convey such Unit free and clear of all liens (other than the lien of current general taxes and the lien of any assessments, charges, or taxes imposed by governmental or quasi-governmental authorities), and Builder' or such Additional Builder's percentage of undivided interest in the Common Area shall have been released therefrom by partial release duly recorded.
- 8.9 <u>Views</u>. Views from a Unit and the Condominium Project are not assured or guaranteed in any way. There is no warranty concerning the preservation of any view or view plane from the Condominium Project and each Owner and Occupant in such Owner's Unit acknowledges and agrees that:
 - (a) Completion of the Condominium Project and the future development of land adjacent to or in the vicinity of the Condominium Project may have a detrimental effect on the views from the Unit and other parts of the Condominium Project; and
 - (b) There are no view easements or rights appurtenant to the Unit or the Condominium Project.

ARTICLE 9 USE RESTRICTIONS

9.1 Rules and Regulations. The Association shall have authority to promulgate and enforce such reasonable rules, regulations and procedures as may be necessary or desirable to aid the Association in carrying out any of its functions or to insure that the Condominium Project is

22

maintained and used in a manner consistent with the interest of the Owners. The Owners shall also comply with all of the rules and regulations set forth in the Master Residential Declaration.

- 9.2 <u>Design Guidelines</u>. Pursuant to the Master Residential Declaration, the Condominium Project is subject to certain "Design Guidelines" more particularly described therein. The Association, as may be delegated to the Condominium Project DRC or other "Reviewer" (as defined in the Master Residential Declaration), shall have the authority to implement and enforce the Design Guidelines for the Project, in the manner described in Article 5 of the Master Residential Declaration.
- 9.3 <u>Design Review Committee</u>: The Association shall have a design review committee for the Project (defined above as the "Condominium Project DRC") consisting of no fewer than three (3) nor more than five (5) individuals, who shall serve and be removed and replaced in the Board's discretion. The Declarant shall initially designate the members of the Condominium Project DRC. Upon the termination of the Declarant's control period defined in Section 5.15, the Management Committee shall appoint the Condominium Project DRC. Condominium Project DRC members need not be Owners or representatives of Owners. The Association may compensate Condominium Project DRC members in such manner and amount, if any, as the Management Committee may determine necessary (which amount shall be included as a Common Expense). Except as otherwise set forth herein, the Condominium Project DRC shall function in the same manner with the rights and obligations as, but subject to, the "Design Review Committee" described in Section 5.2 of the Master Residential Declaration.
- 9.4 <u>Signs</u>. No signs whatsoever (including, but not limited to commercial, political, sale or rental and similar signs) shall be erected or maintained on the Property whether in a window or otherwise without the approval of the Association, except:
 - (a) Such signs as many be required by legal proceedings;
 - (b) One house number identification as originally placed by Declarant, Builder or an Additional Builder;
 - (c) Such signs, the nature, number and location of which have been approved by the Management Committee in advance; and
 - (d) Street identification and traffic directional signs erected on or adjacent to the Condominium Project by Salt Lake County, or any other municipal entity, which signs shall not require prior approval from the Management Committee.

Nothing included herein shall prevent Builder, an Additional Builder or Declarant or their agents and assigns from utilizing reasonable signs, flags, markers and sales devices in furtherance of sales activities until all Units have been sold by Builder or an Additional Builder and such signs of customary and reasonable dimensions as may be displayed on a Unit advertising a Unit for sale or lease.

9.5 <u>Nuisance</u>. No noxious or offensive activity shall be carried on upon the Condominium Project, nor shall any activity which might be or become an annoyance or nuisance to the Owners or Occupants be permitted to interfere with their rights of quiet

enjoyment or increase the rate of any insurance or decrease the value of the Units. No Owner or Occupant shall engage in activity within the Condominium Project in violation of any law, ordinance, statute, rule or regulation of any local, county, state or federal body. In accordance with Section 8.4, nothing included herein shall be construed to prevent Declarant, Builder or any applicable Additional Builder and their respective agents from engaging in all forms of construction and sales activities until all Units have been sold by Declarant, Builder or any such Additional Builder.

- 9.6 <u>Temporary Structures</u>. No structure or building of a temporary character, including a tent or shack, shall be placed upon the Condominium Project or used therein unless the same and its proposed use are approved by the Management Committee. Nothing included herein shall be construed as preventing Builder, any Additional Builder or Declarant from using temporary structures or trailers for construction or sales purposes or engaging in all forms of construction and sales activities within the Condominium Project.
- 9.7 Parking and Use of Open Parking/Visitor Parking. Unless otherwise permitted by the Association, and except for "customary parking" and "temporary parking," as permitted by this Section 9.5, no automobiles or other vehicles of any type (including, without limitation, motorcycles, trailers, campers, vans, recreational vehicles or boats) shall be parked, stored or located within any portion of the Condominium Project, including any Unit, Limited Common Area or Common Area. "Customary parking" shall mean the parking of operable automobiles, motorcycles, small trucks and vans (each of which must not be used for commercial purposes and each of which must not exceed 3/4 ton in size and seven (7) feet in height measured from ground level and eighteen (18) feet in length) within the parking spaces designated as an exclusive Limited Common Area for each respective Unit. "Temporary parking" shall mean the use of designated parking areas within the Condominium Project for parking of operable vehicles belonging to invited guests of the Owners and Occupants including the parking of delivery trucks, service vehicles and other commercial vehicles being used in the furnishing of goods and services to the Owners and Occupants as well as parking of vehicles belonging to or being used by Owners, Occupants and invitees during social engagements and for loading and unloading purposes. The Association may adopt Association Rules relating to the admission and temporary parking of vehicles within the Condominium Project and the use of the visitor parking spaces identified on the Plat, if any, including, without limitation, the right to loan or license the visitor parking spaces in the discretion of the Association, the right to remove or cause to be removed any vehicles that are improperly parked, restrictions on the time visitor spaces may be used and the assessment of charges to Owners and Occupants who violate, or whose invitees violate, such rules. Any charges so assessed shall be special Assessments. Until all the Units are sold, nothing included herein shall be construed to prevent Builder, any Additional Builder or Declarant from using temporary structures or trailers for construction or sales purposes or from engaging in all forms of construction and sales activities within the Condominium Project.
- 9.8 External Fixtures. No external items such as, but not limited to, television and radio antennas, satellite dishes, flag poles, clotheslines, wiring, insulation, air conditioning equipment, water softening equipment, fences, awnings, ornamental screens, screen doors, porch or patio or balcony enclosures, sunshades, lighting fixtures, walls, landscaping and planting, other than those provided in connection with the original construction for the Condominium Project, and any replacements thereof, and other than those approved by the Management Committee, and

any replacements thereof, shall be constructed, erected or maintained on the Condominium Project. The foregoing notwithstanding, until all the Units are sold, nothing included herein shall be construed as preventing Builder, any Additional Builder or Declarant or their agents and assigns from engaging in all forms of construction and sales activities within the Condominium Project.

- 9.9 Window Covers. Each Unit shall have window covers. Only curtains, drapes, shades, shutters and blinds may be installed as window covers, and all such window covers shall be approved in advance by the Management Committee. No window shall be covered by paint, foil, sheets or similar items. The Management Committee may adopt Association Rules regulating the type, color and design of the external surface of window covers.
- 9.10 External Laundering. Unless otherwise permitted by the Management Committee, external laundering and drying of clothing and other items is prohibited.
- 9.11 <u>Outside Speakers and Amplifiers</u>. No radio, stereo, broadcast or loudspeaker units and no amplifiers of any kind shall be placed upon or outside, or be directed to the outside of, any Unit without the prior written approval of the Management Committee.
- 9.12 <u>Repairs</u>. No repairs of any detached machinery, equipment or fixtures, including without limitation, motor vehicles, shall be made upon the Condominium Project.
- 9.13 <u>Unsightly Items</u>. All rubbish, debris or unsightly materials or objects of any kind shall be regularly removed from Units and shall not be allowed to accumulate therein or thereon. Refuse containers and machinery and equipment not a part of the Units, shall be prohibited upon any Unit unless obscured from view of adjoining Units and Common Area. Trash and garbage shall be properly disposed of by Owners and Occupants in refuse containers approved by Salt Lake County or its subcontractor for regularly scheduled pick up and removal, in accordance with the Association Rules applicable thereto adopted by the Management Committee. The Management Committee may adopt rules applicable to the provisions of this Section 9.13 and their enforcement, including, subject to the provisions of Section 5.14 of this Declaration, the assessment of charges to Owners and Occupants who violate, or whose invitees violate, such rules. Any charges so assessed shall be special Assessments. The foregoing notwithstanding, nothing included herein shall be construed as preventing Builder, any Additional Builder or Declarant and their agents and assigns from engaging in all forms of construction and sales activities within the Condominium Project.
- 9.14 Oil and Mineral Activity. No oil drilling, oil development operations, oil refining, quarrying or mining operations of any kind shall be permitted upon the surface of the Condominium Project, nor shall oil wells, tanks, tunnels, mineral excavations or shafts be installed upon the surface of the Condominium Project or within five hundred (500) feet below the surface of the Condominium Project. No derrick or other structure designed for use in boring for water, oil or natural gas shall be erected, maintained or permitted upon the Condominium Project.
- 9.15 Animals. No animals or birds of any kind shall be raised, bred or kept in any Unit, except to the extent permitted by the Master Residential Declaration and any additional rules or

regulations adopted by the Management Committee. Notwithstanding the foregoing, no animal or fowl may be kept within a Unit which, in the good faith judgment of the Management Committee, results in an annoyance or is obnoxious to Owners or Occupants within the Condominium Project and the Management Committee may exercise its judgment for specific pets even though others are permitted to remain. All animals permitted to be kept by this Section 9.15 shall be kept on a leash, except within a Unit, and all fecal matter shall be immediately cleaned up on any portion of the Common Area. The Management Committee may adopt Association Rules applicable to the provisions of this Section 9.15 and to the keeping of pets within the Condominium Project, and their enforcement, including the assessment of charges to Owners and Occupants who violate such rules. Any charges so assessed shall be special Assessments.

- Leases. Any agreement for the leasing, rental, or occupancy of a Unit (a "Lease") shall be in writing and a copy thereof shall be delivered to the Management Committee before the term of the Lease commences. Every Lease shall provide that the terms of such Lease shall be subject in all respects to the provisions of this Declaration, the Bylaws and the Association Rules as well as the Master Residential Declaration. Each Lease shall further provide that any failure by the Occupant thereunder to comply with the terms of the foregoing documents shall be a default under the Lease. If any Lease does not contain the foregoing provisions, such provisions shall nevertheless be deemed to be a part of the Lease and binding on the Owner and Occupant by virtue of their inclusion in this Declaration. No Owner shall be permitted to lease the Owner's Unit for transient or hotel purposes, which shall be defined as rental for any period of less than thirty (30) days. No Owner may lease less than the Owner's entire Unit. Any Owner who shall lease the Owner's Unit shall be responsible for compliance by the Occupant with this Declaration, the Bylaws and the Association Rules. Failure by an Owner to take legal action, including the institution of a forcible entry and detainer proceeding against the Occupant who is in violation of this Declaration, the Bylaws or the Association Rules within ten (10) days after receipt of written demand so to do from the Management Committee, shall entitle the Association, through the Management Committee, to take any and all such action including the institution of proceedings in forcible entry and detainer on behalf of such Owner against the Occupant. Neither the Association nor any agent retained by the Association to manage the Condominium Project shall be liable to the Owner or Occupant for any eviction under this Section 9.16 that is made in good faith. Any expenses incurred by the Association, including attorneys' fees and costs of suit, shall be repaid by such Owner. Failure by such Owner to make such repayment within ten (10) days after receipt of a written demand therefor shall entitle the Management Committee to levy a special Assessment against such Owner and the Owner's Unit for all such expenses incurred by the Association. In the event such special Assessment is not paid within thirty (30) days of its due date, the Management Committee, subject to the provisions of Section 5.14 of this Declaration, may resort to all remedies of the Association for the collection thereof.
- 9.17 <u>Landscape Maintenance</u>. Declarant and the Association shall have the right to maintain all landscaping in the Common Area and the Limited Common Area as specified in Article 4 of this Declaration. The Owner shall maintain all landscaping in exclusive Limited Common Area as specified in Article 4 of this Declaration. Declarant and the Association shall have the right of access to all Common Area of the Condominium Project which are necessary for such landscape maintenance.

- 9.18 <u>Floor Load</u>. There shall be no floor load in excess of the weight for which the Unit or balcony was designed, unless special arrangements are made, and an engineering determination of floor load capacity in the areas of the heavy use is approved in writing by the Management Committee.
- 9.19 Single Family Occupancy. The use of each Unit is restricted to single family occupancy. Except for those activities conducted as part of the marketing and development program for the Condominium Project by Declarant, no industry, business, trade or commercial activities (other than home professional pursuits without employees, public visits or nonresidential storage and mail), or other use of the Unit, shall be conducted, maintained or permitted in any part of a Unit, nor shall any Unit be used or rented for transient, hotel or motel purposes. Single family occupancy is defined as a single housekeeping unit, operated on a nonprofit, noncommercial basis between its occupants, cooking and eating with a common kitchen and dining areas, where all residents are members of a family related by blood, adoption or marriage, except for not more than two (2) additional persons not so related may reside in a Unit.
- No Subdivision of Units or Further Restrictions. No Unit shall be split, subdivided or separated into two (2) or more Units, and no Owner of a Unit shall sell or lease less than all of the Unit. An Owner of two (2) or more adjacent Units may, however, combine those Units to make a single Unit and then separate the single Unit into the original Units for purposes of selling one or more of the Units; provided however, that no such combination or subsequent separation shall be allowed until the Management Committee has approved such combination or separation, in the Management Committee's sole discretion, in writing. No subdivision plat or covenants, conditions or restrictions shall be recorded by any owner or other Person with respect to any Unit unless the Management Committee has first approved the plat or the proposed covenants, conditions or restrictions, such approval to be evidenced by the Association's signature on the final recorded plat or instrument imposing the covenants, conditions or restrictions. Any plat or covenants, conditions, or restrictions recorded in violation of this Section 9.20 shall be absolutely null and void. The Management Committee's review shall be for the purpose of assuring, in the sole and absolute discretion of the Management Committee, that the plat or covenants, conditions and restrictions are consistent and compatible with the overall plan of development of the Property. However, in no event shall the approval of the Management Committee of any plat or covenant, condition or restriction be an abandonment or waiver of any provision of this Declaration. The provisions of this Declaration shall be and remain superior to any such plat or covenants, conditions or restrictions. The foregoing shall not affect Declarant's right to expand the Condominium Project as provided in Article 16 of this Declaration.
- 9.21 <u>Architectural Control</u>. No exterior changes whatsoever shall be commenced, erected, maintained, made or done without the prior written approval of the Management Committee or any committee established by the Management Committee for that purpose. By way of illustration, but not of limitation, the following are considered exterior changes; painting, landscaping, repairs, excavation, patio covers, screens, doors, evaporative coolers, fireplaces, skylights, storage buildings, solar collectors, shade screens, awnings, window coating or tinting, decorative alterations or other work which in any way alters the exterior appearance of the Property. The Management Committee, or committee established by the Management

Committee for that purpose, may designate the design, style, model and manufacturer of any exterior improvement or alteration which is acceptable to the Management Committee. Such designations shall be for the purpose of achieving uniformity of appearance and preservation of property values. No interior structural changes whatsoever shall be commenced, erected, maintained, made or done without the prior written approval of the Management Committee, or any committee established by the Management Committee for that purpose. By way of illustration, but not of limitation, the following are considered interior structural changes: moving, removing, adding, or altering walls, doorways, and the like.

- 9.22 <u>Lighting</u>. Exterior lighting fixtures and walkway and landscaping lights shall be allowed only to the extent approved by the Management Committee.
- 9.23 <u>Association Rules</u>. The Association shall have the power to make and adopt reasonable Association Rules with respect to activities which may be conducted on any part of the Condominium Project. Subject to the provisions of Section 5.14 of this Declaration, the Management Committee's determination as to whether a particular activity being conducted or to be conducted violates or will violate the Association Rules shall be conclusive unless, at a regular or special meeting of the Association, Owners representing a majority of the Allocated Interests in the votes of the Association vote to the contrary.
- 9.24 <u>Variances</u>. The Management Committee may, at its option and in extenuating circumstances, grant variances from the Restrictions set forth in this Article 9 if the Management Committee determines in its discretion: (a) either (i) that a Restriction would create an unreasonable hardship or burden on an Owner or Occupant, or (ii) that a change of circumstances since the recordation of this Declaration has rendered such Restriction obsolete; and (b) that the activity permitted under the variance will not have any substantial adverse effect on the Owners or Occupants of the Condominium Project and is consistent with the high quality of life intended for residents of the Condominium Project.

9.25 <u>Hazardous Substances</u>.

- (a) The Owners shall comply with applicable Environmental Laws (as defined below), and shall not cause or permit the presence, use, disposal, storage or release of any Hazardous Substances (as defined below), on or within the Condominium Project which are not properly controlled, safeguarded and disposed of. The Owners shall not do, nor allow anyone else to do, anything affecting the Condominium Project that is in violation of any Environmental Law. The preceding two sentences shall not apply to the presence, use of storage on the Condominium Project of small quantities of Hazardous Substances that are generally recognized to be appropriate to maintenance of a Unit or the Condominium Project.
- (b) Each Owner shall indemnify, defend and hold the Association and each and every other Owner harmless from and against any and all claims and proceedings (whether brought by private party or governmental agency) for bodily injury, property damage, abatement or remediation, environmental damage or impairment, or any other injury or damage resulting from or relating to any Hazardous Substances located under or upon or migrating into, under, from or through the Condominium Project, which the

Association or the other Owners may incur due to the actions or omissions of an indemnifying Owner. The foregoing indemnity shall apply: (i) whether or not the release of the Hazardous Substances was caused by an indemnifying Owner, a tenant, invitee or otherwise of an indemnifying Owner; and (ii) whether or not the alleged liability is attributable to the handling, storage, generation, transportation or disposal of Hazardous Substances on the Condominium Project. The obligations of each Owner under this Section 9.25 shall survive any subsequent sale by an indemnifying Owner.

(c) As used in this Section 9.25, "Hazardous Substances" are those substances defined as toxic or hazardous substances by Environmental Law and the following substances: gasoline, kerosene, other flammable or toxic petroleum products, toxic pesticides and herbicides, volatile solvents, materials containing asbestos or formaldehyde and radioactive materials. As used in this Section 9.25, "Environmental Law" means federal laws and laws of the jurisdiction where the Condominium Project is located that relate to health, safety or environmental protection.

ARTICLE 10 INSURANCE

- 10.1 <u>Property Insurance</u>. Commencing not later than the date a Unit is conveyed to a Person other than Declarant, the Association shall obtain and maintain the insurance specified in this Declaration; provided, however, the Association shall always comply with the insurance requirements of the Act. All insurance policies shall be issued by generally acceptable insurance carriers.
 - (a) Hazard Insurance. A "master" or "blanket" type policy of property insurance shall be maintained covering the entire Condominium Project, including: the Common Area; all buildings including all Units (other than the interior content thereof); fixtures, building service equipment, personal property and supplies comprising a part of the Common Area or owned by the Association, and which are of a class typically encumbered by First Mortgages held by FNMA or other similar institutional mortgage investors; but excluding land, foundations, excavations, and other items not normally covered by such policies. References herein to a "master" or "blanket" type of policy of property insurance are intended to denote single entity insurance coverage. At a minimum, such "master" or "blanket" policy shall afford protection against loss or damage by fire, by other perils normally covered by the standard extended coverage endorsement, and by all other perils which are customarily covered with respect to projects similar to the Condominium Project in construction, location and use, including, without limitation, all perils normally covered by the standard "all risk" endorsement, where such endorsement is available. Such "master" or "blanket" policy shall be in an amount not less than one hundred percent (100%) of current replacement cost of all elements of the Condominium Project covered by such policy, exclusive of land, foundations, excavation and other items normally excluded from coverage. insurance policy shall include either of the following endorsements to assure full insurable value replacement cost coverage: (1) a Guaranteed Replacement Cost Endorsement (under which the insurer agrees to replace the insurable property regardless of the cost) and, if the policy includes a coinsurance clause, an Agreed Amount

Endorsement (which waives the requirement for coinsurance); or (2) a Replacement Cost Endorsement (under which the insurer agrees to pay up to one hundred percent (100%) of the Property's insurable replacement cost but not more) and, if the policy includes a coinsurance clause, an Agreed Amount Endorsement (which waives the requirement for coinsurance). The maximum deductible amount for such policy covering the Common Area shall be the lesser of Ten Thousand Dollars (\$10,000) or one percent (1%) of the policy face amount. However, for losses related to Units that are covered by such a policy, the deductible related to each Unit shall be One Thousand Dollars (\$1,000). Funds to cover these deductible amounts shall be included in the Association's operating reserve account.

- (b) Flood Insurance. If any part of the Condominium Project is or comes to be situated in a Special Flood Hazard areas as designated on a Flood Insurance Rate Map, a "master" or "blanket" policy of flood insurance shall be maintained covering the Condominium Project, any machinery and equipment that are not part of a building and all Common Area within the Condominium Project ("Insurable Property") in an amount deemed appropriate, but not less than the lesser of: (i) the maximum limit of coverage available under the National Flood Insurance Administration Program for the Insurable Property within any portion of the Condominium Project located within a designated flood hazard areas; or (ii) one hundred percent (100%) of the insurable value of the Insurable Property. The maximum deductible amount for any such policy shall be the lesser of Five Thousand Dollars (\$5,000) or one percent (1%) of the policy face amount.
 - (1) The name of the insured under each policy required to be maintained by the foregoing (a) and (b) shall be the Association for the use and benefit of the individual Owners (the Owners shall be designated by name, if required). Notwithstanding the requirement of the immediately foregoing sentence, each such policy may be issued in the name of an authorized representative of the Association, including any Insurance Trustee with whom the Association has entered into an insurance trust agreement, or any successor to such Insurance Trustee, for the use and benefit of the individual Owners. Loss payable shall be in favor of the Association (or Insurance Trustee), as a trustee for each Owner and each Owner's Lender in the percentage of common ownership. Each Owner and each Owner's Lender, if any, shall be beneficiaries of such policy. Evidence of insurance shall be issued to each Owner and Lender upon request.
 - (2) Each policy required to be maintained by the foregoing items (a) and (b) shall contain the standard mortgage clause, or equivalent endorsement (without contribution), commonly accepted by private institutional mortgage investors in the area in which the Condominium Project is located. If FNMA is a holder of one or more First Mortgages on Units within the Condominium Project, such mortgage clause shall name FNMA or FNMA's servicer of such mortgages as mortgagee. If FNMA's servicer is named as mortgagee in such mortgage clause, such servicer's name shall be followed therein by the phrase "its successors and assigns." In addition, such mortgage clause or another appropriate provision of each such policy shall provide that the policy may not be canceled or substantially modified without at least ten (10) days' prior written notice to the

Association and to each Lender which is listed as a scheduled holder of a mortgage in the policy.

- (3) Each policy required to be maintained by the foregoing items (a) and (b), shall provide, if available, for the following: recognition of any insurance trust agreement; a waiver of the right of subrogation against Owners individually; the insurance is not prejudiced by any act or neglect of individual Owners which is not in the control of such Owners collectively; and the policy is primary in the event the Owner has other insurance covering the same loss.
- (4) Each policy required to be maintained by the foregoing item (a) shall also contain or provide the following: (i) "Inflation Guard Endorsement," if available; (ii) "Building Ordinance or Law Endorsement," if the enforcement of any building, zoning or land use law will result in loss or damage, increased cost of repairs or reconstruction, or additional demolition and removal costs (the endorsement must provide for contingent liability from the operation of building laws, demolition costs and increased costs of reconstruction); and (iii) "Steam Boiler and Machinery Coverage Endorsement," if the project has central heating or cooling, which shall provide that the insurer's minimum liability per accident at least equals the lesser of Two Million Dollars (\$2,000,000) or the insurable value of the building containing the boiler or machinery. In lieu of obtaining this coverage as an endorsement to the commercial package property, the Association may purchase separate stand-alone boiler and machinery coverage.
- 10.2 Comprehensive Public Liability Insurance. To the extent available, the Association shall obtain comprehensive general liability insurance insuring the Association, Declarant, the agents and employees of the Association and Declarant, the Owners and Occupants and the respective family members, guests and invitees of the Owners and Occupants, against liability incident to the use, ownership or maintenance of the Common Area or membership in the Association. The coverage limits under such policy shall be in amounts generally required by private institutional mortgage investors for projects similar to the Condominium Project in construction, location and use. Nevertheless, such coverage shall not be less than One Million Dollars (\$1,000,000) covering all claims for death of or injury to any one person or property damage in any single occurrence. Such insurance shall contain a Severability of Interest Endorsement or equivalent coverage which would preclude the insurer from denying the claim of an Owner because of the negligent acts of the Association or another Owner or Occupant. Such insurance shall also include protection against water damage liability, liability for nonowned and hired automobiles and liability for the property of others. Such insurance must provide that, despite any provisions giving the carrier the right to elect to restore damage in lieu of a cash settlement, such option shall not be exercisable without the approval of the Association. The Management Committee shall adjust the amount of the insurance carried under this Section 10.2 from time to time. The insurance policy carried under this Section 10.2 shall provide that the policy may not be cancelled or substantially modified without at least ten (10) days' prior written notice to the Association and to each Lender.

- 10.3 <u>Workmen's Compensation Insurance</u>. The Management Committee shall purchase and maintain in effect workmen's compensation insurance for all employees of the Association to the extent that such insurance is required by law.
- 10.4 <u>Fidelity Insurance</u>. The Management Committee shall obtain fidelity coverage against dishonest acts on the part of Committee Members, officers, employees or volunteers who handle or who are responsible for handling funds of the Association. Such fidelity bonds shall name the Association as obligee and shall be written in an amount equal to one hundred fifty percent (150%) of the estimated current annual Common Expenses of the Association, including reserves, and shall contain waivers of any defense based on the exclusion of persons who serve without compensation from any definition of "employee" or similar terms or expressions.
- 10.5 <u>Premiums</u>. Premiums upon insurance policies purchased by the Management Committee on behalf of the Association shall be paid by the Association as part of the Common Expenses.

10.6 Policy Provisions.

- (a) Any insurer that has issued an insurance policy to the Association under this Article shall be a generally acceptable insurance carrier according to guidelines set forth in FNMA's Conventional Home Mortgage Selling Contract Supplement or the Federal Home Loan Corporation's Sellers Guide. In addition, any insurer that has issued an insurance policy to the Association under this Article shall issue a certificate or memoranda of insurance to the Association and upon request, to any Owner or Lender.
- (b) The named insured under any policy of insurance shall be the Association, as trustee for the Owners, or its authorized representative, including an Insurance Trustee, who shall have exclusive authority to negotiate losses under the policies. The policy shall provide that each Owner shall be an insured person under the policy with respect to liability arising out of such Owner's interest in the Common Area or membership in the Association. Each Owner hereby appoints the Association, or any Insurance Trustee, as attorney-in-fact for the purpose of purchasing and maintaining insurance required by this Declaration, and adjustment of all losses related thereto, including: the collection and appropriate disposition of all insurance proceeds, the negotiation of all losses and execution of releases of liability, the execution of all documents and the performance of all other acts necessary to administer such insurance. The Association shall receive, hold, or otherwise properly dispose of any proceeds of insurance in trust for the Owners and their Lenders, as their interests may appear. This power-of-attorney is coupled with an interest, shall be irrevocable, and shall be binding on any heirs, personal representatives, successors or assigns of an Owner.
- (c) The Association's insurance shall contain the "Special Condominium Endorsement" or its equivalent. Insurance coverage obtained by the Association shall be primary insurance and may not be brought into contribution with insurance purchased by the Owners.

- (d) Coverage must not be limited by (i) any act or neglect by Owners or Occupants which is not within control of the Association, or (ii) any failure of the Association to comply with any warranty or condition regarding any portion of the Condominium Project over which the Association has no control.
- (e) Coverage may not be canceled or substantially modified (including cancellation for nonpayment of premium) and the insurer may not refuse to renew the policy without at least thirty (30) days' prior written notice to the Association and all Lenders, and to any Owner to whom a certificate has been issued.
- (f) All policies must contain a waiver of subrogation by the insurer as to any and all claims against the Association, the Owners, Occupants and their respective agents and employees, and any defenses based on coinsurance or on invalidity arising from acts of the insured.
- 10.7 <u>Supplemental Insurance</u>. The Management Committee may obtain such other policies of insurance in the name of the Association as the Management Committee deems appropriate to protect the Association and Owners. The Management Committee shall obtain Committee Member's and officer's liability insurance for officers and Committee Members of the Association in accordance with this Declaration. Notwithstanding any of the provisions herein, the Association shall continuously maintain in effect such casualty, flood and liability insurance and a fidelity bond meeting the insurance and fidelity bond requirements for condominium projects established by FNMA, the Department of Housing and Urban Development, the Department of Veterans Affairs and the Government National Mortgage Association, so long as any is a mortgagee or Owner of a Unit, except to the extent such coverage is not available or has been waived in writing by FNMA, the Department of Housing and Urban Development, the Department of Veterans Affairs or the Government National Mortgage Association.
- 10.8 <u>Annual Insurance Report.</u> Not later than sixty (60) days prior to the beginning of each fiscal year, the Management Committee may obtain a written report by a reputable independent insurance broker or consultant setting forth the existing insurance obtained pursuant to this Declaration and stating whether, in the opinion of such broker or consultant, the insurance complies with the requirements of this Declaration and the Act. Such report may also set forth recommendations regarding current policy provisions and for additional insurance reasonably required for the protection of the Owners and the Lenders in light of the insurance then available and the prevailing practice with respect to other similar condominium projects. The Management Committee shall be fully protected in relying on the written report furnished pursuant to this Section 10.8 provided reasonable care and prudence were exercised in selecting such independent insurance broker or consultant.
- 10.9 <u>Insurance Obtained by Owners</u>. Notwithstanding the above, and pursuant to Section 57-8-29 of the Act, an Owner or Occupant shall be permitted to insure the Owner's Unit for the Owner's own benefit.

ARTICLE 11 DESTRUCTION OF IMPROVEMENTS

- 11.1 <u>Reconstruction</u>. In the event of partial or total destruction of a building or buildings or any portion of the Common Area within the Condominium Project, the Management Committee shall promptly take the following actions:
 - (a) The Management Committee shall ascertain the cost of reconstruction by obtaining fixed price bids from at least two (2) reputable contractors, including the obligation to obtain performance and lien payment bonds.
 - (b) The Management Committee shall determine the amount of insurance proceeds, if any, payable by contacting the appropriate representative of the insurer carrying the policy covering the Condominium Project.
 - (c) Pursuant to Section 57-8-30 of the Act, if the insurance proceeds are sufficient to reconstruct the building, such proceeds shall be applied to such reconstruction.
 - (d) If the Management Committee determines: (i) that insurance proceeds will cover eighty-five percent (85%) or more, but not all, of the estimated cost of reconstruction, or (ii) that available insurance proceeds together with available reserves and a Special Assessment equal to twenty-five percent (25%) or less of the then aggregate annual regular Assessments for all Units will completely cover the estimated cost of reconstruction, then the Management Committee shall cause notice to be sent to all Owners and to all Lenders encumbering Units within the Condominium Project setting forth such findings and informing the Owners and Lenders that the Management Committee intends to commence reconstruction pursuant to this Declaration. In the event that Owners representing at least twenty percent (20%) of the Allocated Interests in the votes of the Association object in writing to such reconstruction as indicated in such notice, the Management Committee shall call a special meeting of the Owners pursuant to Section 11.2. In the event that the foregoing requirements are satisfied and the requisite number of Owners do not object in writing to such reconstruction, the Management Committee shall cause reconstruction to take place as promptly as practicable thereafter. In connection with such reconstruction, the Management Committee shall levy a uniform special Assessment against each Owner at such time and in such amount as the Management Committee shall determine is necessary to cover the costs of reconstruction in excess of insurance proceeds and available reserves.
 - (e) If the Management Committee in good faith determines that none of the bids submitted under this Section 11.1 reasonably reflects the anticipated reconstruction costs, the Management Committee shall continue to attempt to obtain an additional bid which it determines reasonably reflects such costs. Such determination shall be made by the Management Committee as soon as possible. However, if such determination cannot be made within ninety (90) days after the date of such destruction because of the unavailability or unacceptability of an insurance estimate or reconstruction bid, or otherwise, the Management Committee shall immediately call a meeting of the affected Owners and all Lenders pursuant to Section 11.2.

- (f) If the Management Committee determines that any Unit is uninhabitable by reason of its total or partial destruction, the Management Committee may abate Assessments against the Owner thereof until the Management Committee determines that habitability has been restored.
- Reconstruction by Vote. If reconstruction is not to take place pursuant to Section 11.1, as soon as practicable after the same has been determined, the Management Committee shall call a special meeting of the Owners by mailing a notice of such meeting to each such Owner. Such meeting shall be held not less than ten (10) days and not more than sixty (60) days after the date of such notice. Unless the Owners, by a vote at such meeting or by the written consent of not less than seventy-five percent (75%) of the Allocated Interests in the votes of the Association (including every Owner of a Unit or an allocated Limited Common Area which will not be rebuilt) determine not to proceed with such reconstruction, reconstruction must take place and the Management Committee shall levy a uniform special Assessment against each Owner at such time and in such amount as the Management Committee shall determine is necessary to cover the costs of reconstruction in excess of insurance proceeds and available reserves.
- 11.3 Procedure for Minor Reconstruction. If the cost of reconstruction is equal to or less than ten percent (10%) of the estimated fair market value of all of the Units in the Condominium Project, then the Management Committee shall contract with a licensed contractor or contractors to rebuild or repair such damaged or destroyed portions of the Condominium Project in conformance with the original plans and specifications, or if the Management Committee determines that adherence to such original plans and specifications is impracticable or is not in conformance with applicable laws, ordinances, building codes or other governmental rules or regulations then in effect, then such repairs or rebuilding shall be of a kind and quality substantially equivalent to the original construction of such improvements.
- <u>Procedure for Major Reconstruction</u>. If the cost of reconstruction is greater than ten percent (10%) of the estimated fair market value of all of the Units in the Condominium Project, all insurance proceeds, together with such amounts from available reserves or special Assessments as are needed to complete the cost of reconstruction, shall be paid directly to an Insurance Trustee, to be designated by the Management Committee, as trustee for all Owners and Lenders. The Insurance Trustee shall be a bank or savings and loan association with an office in Salt Lake County, Utah, whose accounts are insured by the Federal Deposit Insurance Corporation or the successor to such agency. Such proceeds shall be received, held and administered by the Insurance Trustee subject to the provisions of an insurance trust agreement which shall be consistent with the provisions of this Declaration and which shall be entered into between the Insurance Trustee and the Management Committee. Disbursement of such funds shall be made only upon the signatures of two Committee Members and upon the terms and conditions provided in this Section 11.4. As soon as practicable after notification of the receipt of insurance proceeds by the Insurance Trustee, the Management Committee shall enter into a contract with a licensed contractor or contractors for the repair or rebuilding of all of the damaged or destroyed Units and Common Area according to the original plan and specifications of said improvements or, if the Management Committee determines that adherence to such original plans and specifications is impracticable or not in conformity with applicable statutes, ordinances, building codes or other governmental rules and regulations then in effect, then of a quality and kind substantially equivalent to the original construction of such improvements. The

contract with such licensed contractor or contractors shall provide for payment to the contractor or contractors a specified sum for performance and execution of the work therein described, and shall have provisions for periodic disbursement of funds by the Insurance Trustee, which shall be consistent with procedures then followed by prudent lending institutions doing business in Salt Lake County, Utah. Such periodic disbursements of funds shall be for specific dollar amounts and shall not be paid until the contractor who is engaged by the Management Committee shall furnish to the Management Committee before the commencement of construction a full performance and lien payment bond written by a good and responsible corporate surety. Disbursements to the contractor shall be made subject to the prior presentation of an architect's certificate or other documentation containing such provisions as may be appropriate in the circumstances and deemed suitable by the Management Committee. The Management Committee may employ a licensed architect to supervise the repair and rebuilding to insure that all work, services and supplies are in conformity with the requirements of the construction contract.

- 11.5 <u>Determination not to Reconstruct Without Termination</u>. If Owners of not less than seventy-five percent (75%) of the Allocated Interests in the votes of the Association (including every Owner of a Unit or an allocated Limited Common Area which will not be rebuilt after a casualty) and Eligible Mortgagees on Units to which at least fifty-one percent (51%) of the Allocated Interests are attributable vote not to rebuild and the entire Condominium Project is not repaired or replaced, and the Condominium Project is not terminated in accordance with the Act, the insurance proceeds shall be distributed as provided by the Act and the Allocated Interests are automatically reallocated as provided by the Act. In such event, the Association shall promptly prepare, execute and record an amendment to the Declaration reflecting the reallocations.
- 11.6 <u>Negotiations with Insurer</u>. The Association shall have full authority to negotiate in good faith with representatives of the insurer of any totally or partially destroyed building or any other portion of the Common Area, and to make settlements with the insurer for less than full insurance coverage on the damage to such building or any other portion of the Common Area. Any settlement made by the Association in good faith shall be binding upon all Owners and Lenders.
- 11.7 <u>Repair of Units</u>. Installation of improvements to, and repair of any damage to, the interior of a Unit shall be made by and at the individual expense of the Owner of that Unit and, in the event of a determination to reconstruct after partial or total destruction, shall be completed as promptly as practicable and in a lawful and workmanlike manner.
- 11.8 <u>Priority</u>. Nothing contained in this Article shall entitle an Owner to priority over any Lender under a lien encumbering the Owner's Unit as to any portion of insurance proceeds allocated to such Unit.

ARTICLE 12 EMINENT DOMAIN

12.1 <u>Total Taking of a Unit</u>. If a Unit is taken by eminent domain, or sold under threat thereof, or if part of a Unit is taken by eminent domain, or sold under threat thereof, leaving the

LA3:1127855.6 36

Owner with a remnant which may not be practically or lawfully used for any purpose permitted by this Declaration, the award must compensate the Owner for the Owner's Unit and Allocated Interest in the Common Area, regardless of whether any Common Area is taken. Upon such a taking, unless the decree otherwise provides, that Unit's Allocated Interest in the Common Area shall automatically be reallocated to the remaining Units in proportion to their respective interests immediately before the taking. Upon such a taking, the Association shall prepare, execute and record an amendment to the Declaration in compliance with the Act. Any remnant of a Unit remaining after part of a Unit is taken shall become part of the Common Area.

- 12.2 Partial Taking of a Unit. Except as provided in Section 12.1, if part of a Unit is taken by eminent domain, or sold under threat thereof, so that such Unit may still be practically and lawfully used under this Declaration, the award must compensate the Owner for the reduction in the value of the Owner's Unit and Allocated Interest in the Common Area, regardless of whether any Common Area is taken. Upon such a taking, unless the decree otherwise provides, that Unit's Allocated Interest in the Common Area shall remain the same, but if the decree provides for a reduction of the Allocated Interest for such Unit, the reduced amount shall automatically be reallocated to that Unit and the remaining Units in proportion to their respective Allocated Interests immediately before the taking, with the partially acquired Unit participating in the reallocation on the basis of its reduced Allocated Interest.
- 12.3 <u>Taking of Limited Common Area</u>. If the portion of the Condominium Project taken by eminent domain, or sold under threat thereof, is comprised of or includes any Limited Common Area or portion thereof, the portion of the award attributable to the Limited Common Area so taken shall be divided among the Owners of the Units to which such Limited Common Area was allocated at the time of the acquisition.
- 12.4 <u>Taking of Common Area</u>. If the portion of the Condominium Project taken by eminent domain, or sold under threat thereof, is not comprised of or includes any Unit or Limited Common Area, the Management Committee shall, as soon as practicable, cause the award to be utilized for the purpose of repairing or restoring the portion of the Condominium Project so taken, and the portion of the award not used for restoration shall be divided among the owners in proportion to their Allocated Interest in the Common Area before the taking.
- 12.5 <u>Taking of Entire Condominium Project</u>. In the event the Condominium Project, in its entirety, is taken by eminent domain, or sold under threat thereof, the Condominium Project is terminated and the provisions of the Act apply.
- 12.6 Priority and Power of Attorney. Nothing contained in this Article 12 shall entitle an Owner to priority over any Lender under a lien encumbering the Owner's Unit as to any portion of any condemnation award allocated to such Unit. Each Owner hereby appoints the Association as attorney-in-fact for the purpose of negotiations and settlement with the condemning authority for the acquisition of the Common Area, or any part thereof. In the event the taking involves all or part of any Unit or the Common Area or Limited Common Area, the award or proceeds shall be payable to the Association for the use and benefit of the Owners and their Lenders as their interests may appear. This power-of-attorney is coupled with an interest, shall be irrevocable, and shall be binding on any heirs, personal representatives, successors or assigns of an Owner.

ARTICLE 13 RIGHTS OF LENDERS

- 13.1 <u>Notice of Lenders</u>. A Lender shall not be entitled to receive any notice which this Declaration requires the Association to provide Lenders for notice, approval or consent regarding a proposed action or otherwise, unless and until such Lender or its mortgage servicing contractor, has delivered to the Association a written notice stating that such Lender is the holder of a loan encumbering a Unit within the Condominium Project, in accordance with Section 13.6. Notwithstanding the foregoing, if any right of a Lender under this Declaration is conditioned on a specific written request to the Association, in addition to the above, a Lender must also make such request in writing delivered to the Association. Except as provided in this Section 13.1, a Lender's rights pursuant to this Declaration, including, without limitation, the priority of any mortgage lien over the lien of Assessments levied by the Association hereunder shall not be affected by the failure to deliver a notice or request to the Association. The provisions of this Section 13.1 shall not apply to the Department of Veterans Affairs or the Department of Housing and Urban Development.
- 13.2 <u>Priority of Lenders</u>. No breach of the Restrictions herein contained, nor the enforcement of any lien provision herein, shall affect, impair, defeat or render invalid the lien or charge of any Lender made in good faith and for value encumbering any Unit, but all of such Restrictions shall be binding upon and effective against any Owner whose title to a Unit is derived through foreclosure or trustee's sale, or otherwise.

13.3 Relationship with Assessment Liens.

- (a) The lien provided for in Article 6 for the payment of Assessments shall be subordinate to the lien of any Lender which was recorded prior to the date any such Assessment becomes due.
- (b) If any Unit which is subject to a monetary lien created by this Declaration is also subject to the lien of a Lender, then: (i) the foreclosure of any lien created by this Declaration shall not operate to affect or impair the lien of such Lender; and (ii) the foreclosure of the lien of a Lender or the sale under a power of sale included in a mortgage or deed of trust shall not operate to affect or impair the lien hereof, except that any person who obtains an interest thereafter shall take title free of any lien created by this Declaration for any Assessments which became due after the recordation of the mortgage or deed of trust, or any personal obligation for such charges, but such Person shall remain subject to the lien hereof for all charges that shall accrue subsequent to such foreclosure.
- (c) Without limiting the provisions of Section 13.3(b), any Lender who obtains title to a Unit by reason of any foreclosure, or deed or assignment in lieu of foreclosure, or any person who obtains title at a private or judicial foreclosure sale, shall take title to such Unit free of any lien or claim for unpaid Assessments against such Unit which accrued prior to the time such Lender or purchaser takes title to such Unit, except for liens or claims for a share of such Assessments resulting from a pro rata reallocation of such Assessments to all Units within the Condominium Project.

- (d) Nothing in this Section 13.3 shall be construed as releasing any Person from the personal obligation to pay for any Assessments levied pursuant to this Declaration during the period such Person is an Owner.
- 13.4 <u>Required Lender Approval</u>. Except upon the prior written approval of seventy-five percent (75%) of all Lenders which have provided notice to the Association as described in Section 13.1 and Section 13.6, based on one vote for each Unit encumbered by a loan, neither the Association nor the Management Committee shall be entitled by action or inaction to do any of the following:
 - (a) Abandon or terminate by an act or omission the legal status of the Condominium Project; or
 - (b) Except as specifically provided by this Declaration, amend any provisions governing the following:
 - (1) Voting rights;
 - (2) Increases in Assessments that raise the previously assessed amount by more than twenty-five percent (25%), Assessment liens, or the priority of Assessment liens;
 - (3) Reductions in reserves for maintenance, repair and replacement of the Common Area;
 - (4) Reallocation of interests in the Common Area and the Limited Common Area, or rights to their use;
 - (5) Redefinition of any Unit boundaries;
 - (6) Convertibility of Units into Common Area or vice versa;
 - (7) Expansion or contraction of the Condominium Project, or the addition, annexation or withdrawal of property to or from the Condominium Project;
 - (8) Hazard or fidelity insurance requirements;
 - (9) Imposition of any restrictions on the leasing of Units;
 - (10) Imposition of any restrictions on an Owner's right to sell or transfer such Owner's Unit;
 - (11) Restoration or repair of the Condominium Project (after damage or particular condemnation) in a manner other than that specified in this Declaration, the Articles or the Bylaws;
 - (12) Responsibility for maintenance and repair of the general portions of the Condominium Project;

39

- (13) Establishing self-management by the Association where professional management has been required by the Department of Veterans Affairs, the Department of Housing and Urban Development or FNMA; or
- (14) Any provision that expressly benefits Lenders (including their insurers or guarantors).

Any Lender, other than the Department of Veterans Affairs, the Department of Housing and Urban Development or FNMA, who receives, by certified or registered mail, a written request, with a return receipt requested, to approve a change and who does not return a negative response within thirty (30) days shall be deemed to have approved such request.

- 13.5 Other Rights of Lenders. Any Lender (and such Lender's insurer or guarantor) shall, upon written request to the Association, be entitled:
 - (a) To inspect current copies of this Declaration (and any amendments), the Association's Articles, Bylaws, Association Rules and other books and records of the Association during normal business hours; and
 - (b) To receive an annual financial statement of the Association within ninety (90) days following the end of the Association's fiscal year. Upon written request by a Lender, the Association shall furnish the Lender, within a reasonable time, an audited financial statement of the Association for the immediately preceding fiscal year.
- 13.6 <u>Notices of Action</u>. Upon written request to the Association identifying the name and address of the Lender (and the name and address of the Lender's insurer or guarantor, if desired) and the Unit Number or address, any such Lender (or any such insurer or guarantor) will be entitled to timely written notice of:
 - (a) Any condemnation or casualty loss which affects a material portion of the Condominium Project or any Unit on which there is a First Mortgage held by such Lender;
 - (b) Any delinquency in the payment of Assessments or charges owed by an Owner of a Unit subject to the lien of a Lender, which remains uncured for a period of sixty (60) days;
 - (c) Any lapse, cancellation, or material modification of any insurance policy or fidelity bond maintained by the Association; and
 - (d) Any proposed action by the Owners or the Association which would amount to a material change in the Declaration as identified in Section 13.4 of the Declaration.

ARTICLE 14 TERMINATION

14.1 <u>Required Vote</u>. Except as otherwise provided in Article 11 and Article 12, the Condominium Project may be terminated only by unanimous agreement of Owners of all Units.

40

- 14.2 <u>Termination Agreement</u>. An agreement to terminate shall be evidenced by the execution or ratification of a termination agreement, in the same manner as a deed, by the requisite number of Owners. Such an agreement to terminate shall also be approved by unanimous consent of all Lenders with a loan secured by a Unit. Such approval (and any other approval related to an amendment to this Declaration) shall be deemed given when a Lender (except for the Department of Veterans Affairs, the Department of Housing and Urban Development and FNMA) fails to submit a response within thirty (30) days after it receives proper notice of the proposal, provided the notice was delivered by certified or registered mail, with a "return receipt" requested. The termination agreement shall specify a date after which the agreement will be void unless it is recorded before that date. A termination agreement, including all ratifications of such termination agreement, shall be recorded in the records of the County Recorder in Salt Lake County, Utah and is effective only on recordation.
- 14.3 <u>Sale of Condominium Project</u>. A termination agreement may provide that the entire Condominium Project shall be sold following termination. If, pursuant to such agreement, any real estate in the Condominium Project is to be sold following termination, the termination agreement shall set forth the minimum terms of the sale.
- 14.4 <u>Association Duties</u>. The Association, on behalf of the Owners, may contract for the sale of real estate in the Condominium Project, but the contract is not binding on the Owners until approved pursuant to Sections 14.1 and 14.2 of this Declaration. If any real estate in the Condominium Project is to be sold following termination, title to that real estate on termination vests in the Association as trustee for all Owners. Thereafter, the Association has all powers necessary and appropriate to effect the sale. Until the sale has been concluded and the proceeds of the sale distributed, the Association continues in existence with all powers it had before termination. Proceeds of the sale shall be distributed to Owners and Lenders as their interests may appear, based on the relative value of each Unit and the Condominium Project. Unless otherwise specified in the termination agreement, as long as the Association holds title to the real estate, each Owner and their successors in interest have an exclusive right to occupancy of the portion of the real estate that formerly constituted their Unit in accordance with the terms of this Declaration. During the period of that occupancy right, each Owner and their successors in interest remain liable for all Assessments and other obligations imposed on Owners by this Declaration.
- 14.5 <u>Proceeds of Sale</u>. Following termination of the Condominium Project, the proceeds of any sale of real estate, together with the assets of the Association, shall be held by the Association as trustee for Owners and Lenders as their interests may appear. The interest of any Owner in such proceeds shall not be distributed to such Owner except upon the prior payment in full of any Assessment lien or lien of a Lender encumbering such proceeds. Following termination, Lenders holding notes on the Units which were recorded before termination may enforce those liens in the same manner as any lienholder.

ARTICLE 15 AMENDMENTS

15.1 <u>Amendments by Declarant Prior to First Sale</u>. Except as provided elsewhere in this Declaration, prior to the conveyance of the first Unit by Builder or any Additional Builder to

an Owner, this Declaration and any amendments thereto may be amended or revoked by the execution of Declarant of an instrument amending or revoking the same.

- Amendments by Declarant After First Sale. Except as provided elsewhere in this Declaration, Declarant (without obtaining the approval of Owners, Builder, any Additional Builder, the Association, or existing Lenders) may unilaterally amend or modify this Declaration in the exercise of its rights set forth in this Declaration. Also, notwithstanding anything herein to the contrary, during the period in which Declarant controls the Association (as described in Section 5.15), Declarant shall have the unilateral right (without obtaining the approval of the Owners, the Association, Builder, any Additional Builder or existing Lenders) to amend this Declaration, if such amendment is required solely: (a) to comply with applicable law or to correct any error or inconsistency of this Declaration and if such amendment does not adversely affect the rights of any Owner or Lender; or (b) to comply with the rules or guidelines, in effect from time to time, of any governmental or quasi-governmental entity or federal corporation guaranteeing or insuring mortgage loans or governing transactions involving mortgage instruments (including, without limitation, the Department of Veterans Affairs, the Department of Housing and Urban Development, the Federal Housing Administration, FNMA, the Federal Home Loan Mortgage Corporation, the Government National Mortgage Association or any similar agency). If such amendment bears recitation that it is recorded based on such technical error or the requirements of any of the foregoing agencies, such amendment shall not require approval of any Owners or Lenders.
- 15.3 General Amendment Requirements. Except as permitted by Article 3, Section 15.1, Section 15.2, or as otherwise permitted or required by the Act, this Declaration may be amended only by vote or agreement of Owners of Units to which at least sixty-seven percent (67%) of the votes in the Association are allocated. In addition, during the period in which Declarant controls the Association (as described in Section 5.15), this Declaration shall not be amended without Declarant's prior written consent.
- 15.4 <u>Protection of Declarant Rights</u>. An amendment shall not terminate or decrease any unexpired development right, special Declarant right, or period of Declarant control unless the Declarant approves or consents in writing.
- 15.5 Execution of Amendments. An amendment or revocation which only requires the execution of an instrument by Declarant as hereinabove provided shall be effective when executed by Declarant and when recorded in the office of the County Recorder of Salt Lake County, Utah. An amendment which requires the affirmative written assent or vote of the Owners as provided above shall be effective when executed by the President and Secretary of the Association who shall certify that the amendment has been so approved and Declarant if Declarant's consent is also required, and when the amendment has been recorded in the office of the County Recorder of Salt Lake County, Utah.
- 15.6 <u>Lender Approval</u>. Subject to the foregoing, any provision of this Declaration which expressly requires the approval of a specified percentage of the Lenders for action to be taken under such provision can be amended only with the affirmative written assent or vote of not less than the same percentage of the Lenders; provided that in the event approval is requested in writing from a Lender with respect to a proposed amendment and a negative response is not

LA3:1127855.6 42

returned within thirty (30) days following the Lender's receipt of the request, by certified or registered mail, with a return receipt requested, the Lender shall be deemed to have approved the proposed amendment. After the Department of Veterans Affairs or the Department of Housing and Urban Development has approved this Declaration and the Bylaws, the condominium regime documentation may not be amended or merged with a successor condominium regime without the prior written approval of the approving agency.

ARTICLE 16 EXPANSION

- 16.1 Option to Expand. It is anticipated that additions to the Condominium Project will be developed in a series of phases. Accordingly, Declarant hereby reserves, pursuant to Section 57-8-13.6 of the Act, the right and option to expand the Condominium Project (the "Option to Expand") upon the terms and provisions set forth in this Article 16 without the prior consent of the Owners or the Association. The Option to Expand must be exercised by Declarant within seven (7) years after recordation of this Declaration.
- 16.2 <u>Additional Land</u>. Subject to the power granted Declarant below, the land which is subject to the Option to Expand is more particularly described on Exhibit "B."
- 16.3 Order of Exercise. Declarant may add all or any portion of the Additional Land to the Condominium Project without any limitation whatsoever as to the size of the portion or portions of the Additional Land added to the Condominium Project. Declarant may add portions of the Additional Land to the Condominium Project at different times and from time to time as Declarant determines. No assurances are made with regard to which portion of the Additional Land, if any, will be added to the Condominium Project.
- 16.4 <u>Limitations on Expansion</u>. Declarant shall not be restricted in the location of improvements or additional Units on the Additional Land that may be developed, except as may be required by applicable zoning requirements, ordinances or regulations. The maximum number of Units that may be developed on the Additional Land is 570 (resulting in a maximum number of 600 Units in the fully-expanded Condominium Project). The maximum number of Units per acre that may be developed on any portion of the Additional Land added to the Condominium Project is 32. The maximum and minimum number of Units and the related maximum and minimum ownership interests in and to the Common Area are set forth in Section 3.4 of this Declaration.
- 16.5 <u>Use Restrictions</u>. The Units to be located on the Additional Land shall be subject to the same uses as provided in this Declaration. Each Owner of a Unit constructed on any phase of the Additional Land shall have an unrestricted right of ingress and egress to and from the Owner's Unit over and across all Common Area of the Condominium Property. Each Owner of a Unit constructed on any phase of the Additional Land shall have the exclusive right to use and enjoy any Limited Common Area that may be designated for exclusive use by such Owner.
 - 16.6 <u>Nature of Improvements</u>. With respect to improvements on the Additional Land:
 - (a) No assurances are made regarding any structures to be constructed on any portion of the Additional Land, including with respect to the extent to which any such

structures will be compatible with structures on the Property originally within the Condominium Project in terms of the principal materials to be used and architectural style, except that structures constructed on any portion of the Additional Land will be consistent with the initial Condominium Project structures in terms of quality of construction;

- (b) No assurances are made regarding any improvements to be made on any portion of the Additional Land, including with respect to the location, type, or nature of any such improvements, except that improvements constructed on any portion of the Additional Land will be consistent with the initial Condominium Project improvements in terms of quality of construction;
- (c) No assurances are made regarding the nature of any Units to be created on any portion of the Additional Land, including with respect to the extent to which any such Units will be similar to the Units on the Property originally within the Condominium Project; and
- (d) Declarant reserves the right to create facilities and Limited Common Area within any portion of the Additional Land added to the Condominium Project and makes no assurances regarding the types, sizes, or number of such facilities or Limited Common Areas.
- (e) Declarant reserves the right to create within the Common Area of any portion of the Additional Land added to the Condominium Project community, recreational and other facilities for the common use of all Owners and Occupants.
- 16.7 <u>Substantial Completion</u>. All Common Area improvements constructed on and made to the Additional Land shall be substantially completed prior to adding such improvements to the Condominium Project.
- 16.8 <u>Documentation to Expand</u>. In order to add all or any portion of the Additional Land to the Condominium Project, the Declarant shall:
 - (a) Record, with regard to the Additional Land or any portion thereof that is being added to the Condominium Project as Units, Common Area or Limited Common Area, a new or supplemental Plat ("Supplemental Plat") containing the information necessary to comply with the Act, and showing the location and dimensions of the vertical and horizontal boundaries of each Unit, the Common Area or Limited Common Area, if any, formed out of the Additional Land or a portion thereof, and assigning any Limited Common Area which shall become appurtenant to any such Unit; and
 - (b) Record simultaneously with each Supplemental Plat a Supplemental Declaration, duly executed and acknowledged by the Declarant and all of the owners and lessees of the Additional Land added to the Condominium Project. Each such Supplemental Declaration and Supplemental Plat shall contain a legal description by metes and bounds of that portion of the Additional Land being added to the Condominium Project.

44

The ownership interest in the Common Area for all Units in the Condominium Project, the voting interests for Owners, and the apportionment of the Common Expenses shall change at the time Declarant records a Supplemental Declaration and a Supplemental Plat reflecting Declarant's exercise of the Option to Expand. At such time, each Owner's (i) percentage of Allocated Interest in and to the Common Area, (ii) percentage of votes for all matters of the Association, and (iii) percentage of Allocated Interest in the Common Expenses shall be equal to the percentage obtained by dividing 1 by the total number of Units then contained in the Condominium Project. It is contemplated that there may be multiple amendments filed by Declarant and such amendments and supplements are hereby expressly authorized.

- 16.9 <u>Title to Units</u>. Each Owner, by execution of a contract for deed or the acceptance of a deed to a Unit in the Condominium Project, shall be deemed to have consented to all provisions of this Article 16, including the procedure for adjustment of Unit ownership interests. After the filing for record of any Supplemental Declaration and the Supplemental Plat reflecting Declarant's exercise of the Option to Expand, or any part thereof, legal and equitable title to each Unit thereby created within the Additional Land including the interest in the Common Area appurtenant to each such Unit shall be vested in and held by Declarant.
- 16.10 Conveyance of Additional Land. Declarant shall have the right to convey or otherwise transfer its interest in the Additional Land or any portion thereof and to designate the grantee thereof as an "Additional Builder" hereunder with respect to such portion of the Additional Land in a Supplemental Declaration. Builder may be designated as an Additional Builder hereunder. Such conveyance or transfer shall not terminate this Option to Expand with respect to any portion of the Additional Land and the grantee of any portion of the Additional Land shall have the same option and rights as Declarant under this Declaration.
- 16.11 <u>Amendment</u>. No provision of this Article 16 shall be amended without the prior written consent of Declarant, so long as Declarant either owns or has the right to acquire or construct any Units in the Condominium Project.
- 16.12 <u>Lender Approvals</u>. If the Department of Veterans Affairs, the Department of Housing and Urban Development or FNMA holds, insures or guarantees a mortgage or mortgages secured by a Unit or Units in the Condominium Project, no Additional Land may be added to the Project without the prior written consent of such agency or corporation.

ARTICLE 17 GENERAL PROVISIONS

- 17.1 Enforcement. The Association or any Owner shall have the right to enforce, by proceedings at law or in equity, all Restrictions and other provisions now or hereafter imposed by this Declaration, or any amendments thereto, including the right to prevent the violation of any such Restrictions, and the right to recover damages and other sums for such violation. The Association or any Owner shall also have the right to enforce by proceedings at law or in equity the provisions of the Bylaws and the Association Rules and any respective amendments thereto.
- 17.2 <u>No Waiver</u>. Failure by the Association or by any Owner to enforce any Restriction or provision herein contained, or contained in the Bylaws or the Association Rules, in

45

any certain instance or on any particular occasion shall not be deemed a waiver of such right of enforcement as to any such future breach of the same or any other Restriction or provision.

- 17.3 <u>Cumulative Remedies</u>. All rights, options and remedies of Declarant, the Association, the Owners or the Lenders under this Declaration are cumulative, and no one of them shall be exclusive of any other, and Declarant, the Association, the Owners and the Lenders shall have the right to pursue any one or all of such rights, options and remedies or any other remedy or relief which may be provided by law, whether or not stated in this Declaration.
- 17.4 <u>Severability</u>. Invalidation of any one or a portion of the Restrictions or provisions set forth in this Declaration or in the Bylaws or the Association Rules by judgment or court order shall in no way affect any other Restrictions or provisions contained herein or therein which shall remain in full force and effect.
- 17.5 Covenants to Run with the Land: Term. The Restrictions and other provisions of this Declaration shall run with and bind the Condominium Project as equitable servitudes and also as covenants running with the land and shall inure to the benefit of and shall be enforceable by the Association or any Owner, their respective legal representatives, heirs, successors and assigns for a term of fifty (50) years from the date this Declaration is recorded, after which time the Restrictions and other provisions shall be automatically extended for successive periods of ten (10) years, unless an instrument, signed and acknowledged by Owners of not less than seventy-five percent (75%) of the Allocated Interests in the votes of the Association and their Lenders, has been recorded prior to the end of any such period agreeing to change or revoke the Restrictions and other provisions of this Declaration in whole or in part. By acquiring any interest in a Unit, or in a Limited Common Area, or in the Common Area, such Owner consents to, and agrees to be bound by, each and every provision of this Declaration.
- 17.6 Allocation upon Termination. Unless provided otherwise herein, upon any liquidation or termination of all or part of the Condominium Project, the Association shall represent the Owners in any proceedings, negotiations, settlements or agreements related thereto. Each Owner hereby appoints the Association as attorney-in-fact for such purpose, including the allocation of any losses, awards or proceeds resulting from such termination or liquidation. Any proceeds generated by such a termination or liquidation shall be made payable to the Association, which will hold such proceeds for the benefit of the Owners and their Lenders. Owners shall share in the proceeds of such termination or liquidation in the same proportion as their Allocated Interest in the Common Area (or as otherwise provided by the Act), but in such event, the liens and provisions of all lenders or Assessment liens encumbering Units within the Condominium Project so encumbered shall extend to each applicable Owner's Interest in such proceeds. The interest of an Owner in such proceeds shall not be distributed to such Owner except upon the prior payment in full of any Assessment lien or lien of a Lender encumbering such proceeds.
- 17.7 <u>Construction</u>. The provisions of this Declaration shall be liberally construed to effectuate its purpose of creating a plan for the development of a condominium community and for the maintenance of the Condominium Project. The Article and Section headings have been inserted for convenience only, and shall not be considered or referred to in resolving questions of

46

interpretation or construction. References in this Declaration to Articles and Section numbers, unless otherwise expressly provided, are to the Articles and Sections of this Declaration.

- 17.8 <u>Gender and Number</u>. Whenever the context of this Declaration requires, the singular shall include the plural, and vice versa, and the masculine shall include the feminine and the neuter, and vice versa.
- 17.9 <u>Nuisance</u>. The result of every act or omission whereby any provision or Restriction contained in this Declaration or any provision contained in the Bylaws or the Association Rules is violated in whole or in part is hereby declared to be and shall constitute a nuisance, and every remedy allowed at law or in equity against a nuisance, either public or private, shall be applicable with respect to the abatement thereof and may be exercised by the Association or any Owner. Such remedy shall be deemed cumulative to all other remedies set forth in this Declaration and shall not be deemed exclusive.
- 17.10 Attorneys' Fees. In the event any action is instituted to enforce any of the provisions contained in this Declaration, the Bylaws or the Association Rules, the party prevailing in such action shall be entitled to recover from the other party thereto, as part of the judgment, reasonable attorneys' fees and costs of suit.
- 17.11 <u>Notices</u>. Any notice to be given to an Owner, a Lender or the Association under the provisions of this Declaration shall be in writing and shall be delivered as follows:
 - (a) Notice to an Owner shall be delivered personally or placed in the first class United States mail, postage prepaid, to the most recent address furnished by such Owner in writing to the Association for the purpose of giving notice, or if no such address shall have been furnished, then to the street address of such Owner's Unit. Any notice so deposited in the mail shall be deemed delivered seventy-two (72) hours after such deposit. In the case of co-Owners, any such notice may be delivered or sent to any one of the co-Owners on behalf of all co-Owners and shall be deemed delivered to all such co-Owners.
 - (b) Notice to a Lender shall be delivered by first class United States mail, postage prepaid, to the most recent address furnished by such Lender in writing to the Association for the purpose of notice or, if no such address shall have been furnished, to any office of the Lender in Salt Lake County, Utah, or if no such office is located in Salt Lake County, to any office of such Lender. Any notice so deposited in the mail shall be deemed delivered seventy-two (72) hours after such deposit.
 - (c) The declaration of an officer or authorized agent of the Association declaring under penalty of perjury that a notice has been mailed to any Owner or Owners, or to any Lender or Lenders, to the address or addresses for the giving of notice pursuant to this Section 17.11, shall be deemed conclusive proof of such mailing.
 - (d) Notice to the Association shall be delivered by registered or certified United States mail, postage prepaid, addressed to the office of the statutory agent of the Association or as follows:

Daybreak Eastlake Village Condominiums Owners' Association 5295 South 300 West, Suite 475 Murray, Utah 84107

The Association shall, however, have the right to designate a successor or substitute address for receipt of notices hereunder by filing a Supplemental Declaration.

- 17.12 Effect of Declaration. This Declaration is made for the purposes set forth in the recitals in this Declaration and Declarant makes no warranties or representations, express or implied, as to the binding effect or enforceability of all or any portion of this Declaration, or as to the compliance of any of these provisions with public laws, ordinances, regulations and the like applicable thereto. Declarant shall have no liability whatsoever if any of the provisions of this Declaration, the Bylaws or the Association Rules are determined to be unenforceable in whole or in part or under certain circumstances.
- 17.13 <u>Personal Covenant</u>. To the extent the acceptance of a conveyance of a Unit creates a personal covenant between the Owner of such Unit and Declarant, other Owners or the Association, such personal covenant shall terminate and be of no further force or effect from and after the date when a Person ceases to be an Owner except to the extent this Declaration provides for personal liability with respect to the Assessments incurred during the period a Person is an Owner.
- 17.14 <u>Nonliability of Officials</u>. To the fullest extent permitted by law, neither the Management Committee nor any officer of the Association shall be liable to any Owner or the Association for any damage, loss or prejudice suffered or claimed on account of any decision, approval or disapproval, course of action, act, omission, error or negligence if such Committee Member or officer acted in good faith within the scope of such Person's duties.
- 17.15 <u>Use of Funds Collected by the Association</u>. All funds collected by the Association, including Assessments and contributions to the Association paid by the Owners, if any, shall be held by the Association in a fiduciary capacity to be expended in their entirety for nonprofit purposes of the Association in managing, maintaining, caring for and preserving the Common Area and for other permitted purposes as set forth in this Declaration. No part of said funds shall inure to the benefit of any Owner (other than as a result of the Association managing, maintaining, caring for and preserving the Common Area and other than as a result of expenditures made for other permitted purposes as set forth in this Declaration).
- 17.16 Notification of Sale and Transfer Fee. Concurrently with the consummation of the sale or other transfer of any Unit, or within fourteen (14) days after the date of such transfer, the transferee shall notify the Association in writing of such transfer and shall accompany such written notice with any nonrefundable transfer fee payable pursuant to the Association Rules, to cover Association documentation and processing. The Management Committee may establish a transfer fee, from time to time, which shall be no more than the amount of the then current regular monthly Assessment. The written notice shall set forth the name of the transferee and the transferor, the street address of the Unit purchased or acquired by the transferee, the transferee's mailing address, the date of the sale or transfer and the name and address of the transferee's Lender, if any. Prior to the receipt of such written notice, all notices required or permitted to be

given by the Association to the Owner shall be deemed to be duly made or given to the transferee if duly and timely made and given to the transferee's predecessor in interest. The transfer fee shall be the personal obligation of the new Owner and shall be secured by the lien in Section 6.1. Notwithstanding the other provisions of this Declaration, this Section 17.16 shall not apply to a Lender who becomes an Owner by a foreclosure proceeding or any deed or assignment in lieu of foreclosure.

17.17 Owner Liability and Indemnification. Subject to the provisions of Section 5.14 of this Declaration, each Owner shall be liable to the remaining Owners and to the Association for any damage to the Common Area that may be sustained by reason of the negligence of that Owner or such Owner's family members, tenants, guests or invitees, but only to the extent that any such damage is not covered by casualty insurance in favor of the Association. Each Owner, by acceptance of a deed for a Unit, agrees personally and for family members, tenants, guests and invitees to indemnify each and every other Owner, and to hold such other Owners harmless from, and to defend such Owners against, any claim of any person for personal injury or property damage occurring within the Unit of that particular Owner, including Limited Common Area, if any, except to the extent: (a) that such injury or damage is covered by liability insurance in favor of the Association or any other Owner; or (b) the injury or damage occurred by reason of the willful or negligent act or omission of the Association or other Owner or other Person temporarily visiting such Unit. To the fullest extent permitted by law, the Association shall indemnify the following persons and entities against all expenses and liabilities including, but not limited to, attorneys' fees, reasonably incurred by or imposed upon such person in connection with any proceeding to which such person may be a party, or in which such person may become involved, by reason of such person being or having served in any capacity on behalf of the Association (or by reason of having appointed, removed or controlled or failed to control members of the Management Committee or the Project DRC): (a) every director and officer of the Association and every member of the Project DRC and any other committee of the Association; (b) every director, officer and employee of the Declarant and the Master Residential Association and the Daybreak Community Council, Inc.; and (c) every person serving as an employee of the Association. Any such person shall be entitled to indemnification whether or not such person is a director, officer or member of the Association or of the Condominium Project DRC or any other committee of the Association or is serving in any other such specified capacity at the time such expenses are incurred, provided that the Management Committee shall determine, in good faith, that the person to be indemnified hereunder did not act, fail to act, or refuse to act with gross negligence or fraudulent or criminal intent in the performance of its duties. The foregoing rights of indemnification shall be in addition to, and not in place of, all other rights to which such persons may be entitled at law or otherwise.

17.18 <u>Conflicting Provisions</u>. In the case of any conflict between this Declaration and the Bylaws, or the Association Rules, this Declaration shall control. In the case of any conflict between this Declaration and the Act, to the extent the Act does not legally allow this Declaration to contain provisions contrary to the Act, the Act shall control and this Declaration shall be deemed modified accordingly. Notwithstanding the above, this Declaration shall be deemed modified and amended only to the extent necessary to come into compliance with the Act.

- 17.19 Cooperation with Declarant. Declarant or its affiliate is the owner of real property adjacent to or nearby the Project. The plans for such real property currently include development as mixed-use parcels. Different uses may be constructed at different times. Therefore Lots within the Project may be located near other uses when such adjacent or nearby real property is developed, including retail, civic and office property and apartments, townhouses and condominiums. Certain traits of each Unit and the Project may change over time depending on adjacent or nearby uses, such as the view, shade, perceived privacy and amount of traffic. The Association and each Owner hereby acknowledge the plans for adjacent mixed-use development and agree not to challenge or oppose such development with any of the Association, the Master Residential Association, the Daybreak Community Council, the City of South Jordan or the County of Salt Lake, State of Utah, or any other relevant governing body.
- 17.20 Consent, Power of Attorney, Waiver. By acceptance of a deed, lease or other conveyance of an interest in Unit, each Owner or Occupant of such Owner's Unit consents to the rights reserved to Declarant and the Association in this Declaration, including but not limited to, the right to prepare, execute, file, process and record necessary and appropriate documents and other items to establish and grant easements and to make necessary and appropriate amendments of this Declaration, the Plat and the Bylaws. By such acceptance, each Owner or Occupant agrees to execute all documents and to do all other things as may be necessary or convenient to effect the same; and such acceptance shall be deemed an appointment of Declarant, during the period in which Declarant controls the Association (as described in Section 5.15), and the Association thereafter, with full right of substitution, as the attorney-in-fact of such Owner or Occupant to execute such documents and to do such things on such Owner's or Occupant's behalf; and such appointment, being coupled with an interest, shall be irrevocable for the specific period of Declarant's and the Association's reserved rights as set forth in this Declaration and shall not be affected by the disability of any such Owner or Occupant.
- 17.21 <u>Master Residential Declaration's Full Force and Effect</u>. The Master Residential Declaration shall remain in full force and effect, provided, however, that in the event of a specific and clear conflict between the Master Residential Declaration and this Declaration, the terms of this Declaration shall control, but only in connection with the Condominium Project and the Lots.
- 17.22 Security. The Association, Declarant, KLRDC, any Additional Builder, or Builder shall in no way be considered insurers or guarantors of security within or relating to the Condominium Project, including any Common Area in which the Association may have an obligation to maintain, and the Association, Declarant, KLRDC, any Additional Builder, and Builder shall not be held liable for any loss or damage by reason of any failure to provide adequate security or any ineffectiveness of security measures undertaken. Each Owner or Occupant acknowledges and understands that Declarant, KLRDC, Builder, any Additional Builder, the Association and the Management Committee are not insurers of the safety or well being of Owners or Occupants or of their personal property, and that each Owner or Occupant assumes all risks for loss or damage to persons, the Units, the Common Area, and to the contents of improvements located thereon to the extent not insured by the Association pursuant to Article 10 above. EACH OWNER AND OCCUPANT UNDERSTANDS AND ACKNOWLEDGES THAT DECLARANT, KLRDC, BUILDER, ANY ADDITIONAL BUILDER, THE ASSOCIATION AND THE MANAGEMENT COMMITTEE HAVE NOT MADE ANY

REPRESENTATIONS OR WARRANTIES OF ANY KIND AND THAT EACH OWNER OR OCCUPANT HAS NOT RELIED UPON ANY REPRESENTATIONS OR WARRANTIES, EXPRESSED OR IMPLIED, INCLUDING ANY WARRANTY OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE, RELATIVE TO THE SECURITY OF THE CONDOMINIUM PROJECT.

[Signatures on the Next Page]

IN WITNESS WHEREOF, the undersigned Declarant has executed this Declaration, and KLRDC and Builder have consented to the same, to be effective as of the day and year first above written.

n	ec'	l۵	ra	n	t	•

KENNECOTT LAND COMPANY, a Delaware corporation

Name: M. Bruce Snyde

Title: Vice President Residential and

Commercial Development

STATE OF UTAH)
) SS.
COUNTY OF SALT LAKE)

On July 30, 2007 personally appeared before me, a Notary Public, M. Bruce wider, the We President Residential? Communical Development KENNECOTT LAND COMPANY personally known or proved to me to be the person whose name is subscribed to the above instrument who acknowledged to me that he executed the above instrument on behalf of KENNECOTT LAND COMPANY.

WITNESS my hand and official Seal.



Notary Public in and for said State

My commission expires:

CONSENTED TO BY:

"KLRDC":

KENNECOTT LAND RESIDENTIAL DEVELOPMENT COMPANY, a Delaware corporation

Name: M. Bruce Snyder

Title: Vice President Residential and

Commercial Development

"BUILDER"

HOLMES HOMES, INC., a Utah corporation

Name ___

Title__

. Holmen

S-2

STATE OF UTAH)
COUNTY OF SALT LAKE) SS.)
KENNECOTT LAND RESIDE known or proved to me to be the per who acknowledged to me that	, 2007 personally appeared before me, a Notary Public, of NTIAL DEVELOPMENT COMPANY personally erson whose name is subscribed to the above instrument he executed the above instrument on behalf of TIAL DEVELOPMENT COMPANY.
	WITNESS my hand and official Seal.
WENDY MAURER NOTARY PUBLIC-STATE OF UTAH 9 9651 S. 1075 W. APT. C-303 WEST JORDAN, UT 84088 COMM. EXP. 01-09-2011	Notary Publicin and for said State My commission expires:
STATE OF UTAH)) SS.
COUNTY OF SALT LAKE)
HOLMES HOMES, INC. persona	, 2007 personally appeared before me, a Notary the of lly known or proved to me to be the person whose name nt who acknowledged to me that he executed the above HOMES, INC.

VICKI L FLEMING
MOTARY PUBLIC • STATE OF UTAH
13717 8 HACKAMORE CIRCLE
DRAPER, UT 84020
COMM. EXP. 02/25/2009

My commission expires:

WITNESS my hand and official Seal.

[SEAL]

S-3

Part of Tax Parcel No. 27-18-400-005-0000

EXHIBIT "A"

LEGAL DESCRIPTION PARCEL

The following described real property, located in Salt Lake County, Utah, is the Parcel referred to in the Declaration:

Boundary Description - Eastlake Village Condominiums

PARCEL G

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 2104.505 feet; thence South 89°52'45" West perpendicular to said section line for 3624.215 feet to the POINT OF BEGINNING; thence North 90°00'00" West for 85.000 feet; thence North 00°00'00" East for 86.945 feet; thence North 90°00'00" East for 85.000 feet; thence South 00°00'00" West for 86.945 feet to the POINT OF BEGINNING.

Containing 7,390 sq. ft. or 0.1697 acres

PARCEL H

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 2013.033 feet; thence South 89°52'45" West perpendicular to said section line for 3624.408 feet to the POINT OF BEGINNING; thence North 90°00'00" West for 85.000 feet; thence North 00°00'00" East for 91.472 feet; thence North 90°00'00" East for 85.000 feet; thence South 00°00'00" West for 91.472 feet to the POINT OF BEGINNING.

Containing 7,775 sq. ft. or 0.1785 acres

PARCEL I

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North $00^{\circ}07'15"$ West -5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North $00^{\circ}07'15"$ West along the east line of Section 18 for 1921.561 feet; thence South $89^{\circ}52'45"$ West perpendicular to said section line for 3624.600 feet to the POINT OF BEGINNING; thence North $90^{\circ}00'00"$ West for 85.000 feet; thence North $90^{\circ}00'00"$ East for 91.472 feet; thence North $90^{\circ}00'00"$ East for 85.000 feet; thence South $90^{\circ}00'00"$ West for 91.472 feet to the POINT OF BEGINNING.

Containing 7,775 sq. ft. or 0.1785 acres

LA3:1127855.6 A-1

PARCEL J

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 1831.450 feet; thence South 89°52'45" West perpendicular to said section line for 3624.790 feet to the POINT OF BEGINNING; thence North 90°00'00" West for 85.000 feet; thence North 00°00'00" East for 90.111 feet; thence North 90°00'00" East for 85.000 feet; thence South 00°00'00" West for 90.111 feet to the POINT OF BEGINNING.

Containing 7,659 sq. ft. or 0.1758 acres

PARCEL K

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 1831.416 feet; thence South 89°52'45" West perpendicular to said section line for 3608.790 feet to the POINT OF BEGINNING; thence South 00°00'00" West for 100.000 feet; thence North 90°00'00" West for 101.000 feet; thence North 90°00'00" East for 101.000 feet to the POINT OF BEGINNING.

Containing 10,100 sq. ft. or 0.2319 acres

PARCEL L

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 1830.370 feet; thence South 89°52'45" West perpendicular to said section line for 3112.045 feet to the POINT OF BEGINNING; thence with a non-tangent curve to the left having a radius of 1482.000 feet, whose center bears North 87°13'05" East, with a central angle of 03°23'37" (chord bearing and distance of South 04°28'43" East – 87.766 feet) for an arc distance of 87.778 feet; thence South 82°42'12" West for 81.155 feet; thence with a curve to the right having a radius of 271.000 feet, with a central angle of 05°57'39" (chord bearing and distance of South 85°41'02" West – 28.181 feet) for an arc distance of 28.194 feet; thence North 00°00'00" East for 99.926 feet; thence North 90°00'00" East for 101.746 feet to the POINT OF BEGINNING.

Containing 9,906 sq. ft. or 0.2274 acres

PARCEL M

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North

LA3:1127855.6 A-2

00°07'15" West along the east line of Section 18 for 1920.484 feet; thence South 89°52'45" West perpendicular to said section line for 3113.602 feet to the POINT OF BEGINNING; thence South 00°00'00" West for 18.184 feet; thence with a curve to the left having a radius of 1482.000 feet, with a central angle of 02°46'55" (chord bearing and distance of South 01°23'27" East – 71.950 feet) for an arc distance of 71.957 feet; thence South 90°00'00" West for 86.746 feet; thence North 00°00'00" East for 90.111 feet; thence North 90°00'00" East for 85.000 feet to the POINT OF BEGINNING.

Containing 7,701 sq. ft. or 0.1768 acres

PARCEL N

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 2011.956 feet; thence South 89°52'45" West perpendicular to said section line for 3113.409 feet to the POINT OF BEGINNING; thence South 00°00'00" West for 91.472 feet; thence South 90°00'00" West for 85.000 feet; thence North 00°00'00" East for 91.472 feet; thence South 90°00'00" East for 85.000 feet to the POINT OF BEGINNING.

Containing 7,775 sq. ft. or 0.1785 acres

PARCEL O

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 2103.429 feet; thence South 89°52'45" West perpendicular to said section line for 3113.216 feet to the POINT OF BEGINNING; thence South 00°00'00" West for 91.472 feet; thence North 90°00'00" West for 85.000 feet; thence North 00°00'00" East for 91.472 feet; thence North 90°00'00" East for 85.000 feet to the POINT OF BEGINNING.

Containing 7,775 sq. ft. or 0.1785 acres

PARCEL P

Commencing at the Southeast corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian, (Basis of Bearing is North 00°07'15" West – 5292.036 feet between the Southeast corner and the Northeast Corner of said Section 18) and running thence North 00°07'15" West along the east line of Section 18 for 2190.373 feet; thence South 89°52'45" West perpendicular to said section line for 3113.033 feet to the POINT OF BEGINNING; thence South 00°00'00" West for 86.945 feet; thence South 90°00'00" West for 85.000 feet; thence North 00°00'00" East for 86.945 feet; thence South 90°00'00" East for 85.000 feet to the POINT OF BEGINNING.

Containing 7,390 sq. ft. or 0.1697 acres

LA3:1127855.6 A-3

EXHIBIT "B"

LEGAL DESCRIPTION ADDITIONAL LAND

The following described real property, located in Salt Lake County, Utah, is the Additional Land:

Boundary Description - Additional Land Area

PARCEL A

Commencing at the North Quarter corner of Section 18, Township 3 South, Range 1 West, Salt Lake Base and Meridian (Basis of Bearing North 89°58'34" East - 2639.700 feet between the North Quarter and the Northeast Corner of said Section 18) and running South 89°57'12" West along the north line of Section 18 for 843.482 feet; thence South 00°02'48" West perpendicular to said section line for 945.122 feet to the POINT OF BEGINNING; thence with a non tangent curve to the left having a radius of 429.000 feet, whose center bears North 89°08'36" East, with a central angle of 57°05'30" (chord bearing and distance of South 29°24'09" East – 410.005 feet) for an arc distance of 427.472 feet; thence South 57°56'54" East for 61.183 feet; thence with a curve to the right having a radius of 6.500 feet, a central angle of 90°00'00" (chord bearing and distance of South 12°56'54" East - 9.192 feet) for an arc length of 10.210 feet; thence South 32°03'06" West for 213.500 feet; thence North 57°56'54" West for 276.417 feet; thence with a non-tangent curve to the left having a radius of 160.000 feet, whose center bears South 86°30'47" West, with a central angle of 74°58'34" (chord bearing and distance of North 40°58'30" West - 194.751 feet) for an arc length of 209.373 feet; thence North 00°00'00" East for 285.851 feet; thence North 90°00'00" East for 220.082 feet to the POINT OF BEGINNING.

Containing 3.3281 acres

October 31, 2006

LA3:1127855.6 B-1

EXHIBIT "C"

FORM BYLAWS

[Attached]

LA3:1127855.6 C-1

BK 9498 PG 1404

BYLAWS

OF

DAYBREAK EASTLAKE VILLAGE CONDOMINIUMS OWNERS' ASSOCIATION

A Utah Nonprofit Corporation

1

LA3:1130245.2

BK 9498 PG 1405

TABLE OF CONTENTS

	ra e e e e e e e e e e e e e e e e e e e	ıge
ARTICLE I	NAME AND PRINCIPAL OFFICE	1
1.1	Name	1
1.2	Offices	1
ARTICLE II	DEFINITIONS	1
2.1	Definitions	1
ARTICLE III	MEMBERS	1
3.1	Annual Meetings	1
3.2	Special Meetings	1
3.3	Place of Meetings	1
3.4	Notice of Meetings	2
3.5	Members of Record	2
3.6	Quorum	2
3.7	Proxies	2
3.8	Votes	3
3.9	Waiver of Irregularities	3
3.10	Informal Action by Members	3
ARTICLE IV	MANAGEMENT COMMITTEE	3
4.1	General Powers	3
4.2	Number, Tenure and Qualifications	3
4.4	Regular Meetings	4
4.5	Special Meetings	4
4.6	Quorum and Manner of Acting	5
4.7	Compensation	5
4.8	Resignation and Removal	5
4.9	Vacancies and Newly Created Committee Memberships	5
4.10	Informal Action by Committee Members	5
ARTICLE V	OFFICERS	6
5.1	Officers	6
5.2	Election, Tenure and Qualifications	6
5.3	Subordinate Officers	6

TABLE OF CONTENTS

(continued)

		rage
5.4	Resignation and Removal	6
5.5	Vacancies and Newly Created Offices	6
5.6	The President	6
5.7	The Secretary	6
5.8	The Treasurer	7
5.9	Compensation	7
ARTICLE VI	COMMITTEES	7
6.1	Designation of Committees	7
6.2	Proceeding of Committees	7
6.3	Quorum and Manner of Acting	7
6.4	Resignation and Removal	7
6.5	Vacancies	8
ARTICLE VI	II INDEMNIFICATION	8
7.1	Indemnification	8
7.2	Other Indemnification	8
7.3	Insurance	9
7.4	Settlement by Association	9
ARTICLE VI	III FISCAL YEAR AND SEAL	9
8.1	Fiscal Year	9
8.2	Seal	9
ARTICLE IX	RULES AND REGULATIONS	9
9.1	Rules and Regulations	9
ARTICLE X	AMENDMENTS	9
10.1	Amendments	9

BYLAWS OF

DAYBREAK EASTLAKE VILLAGE CONDOMINIUMS OWNERS' ASSOCIATION A Utah Nonprofit Corporation

Pursuant to the provisions of the Utah Revised Nonprofit Corporation Act (the "Act"), the Management Committee of Daybreak Eastlake Village Condominiums Owners' Association, a Utah nonprofit corporation, hereby adopts the following Bylaws for such nonprofit corporation.

ARTICLE I NAME AND PRINCIPAL OFFICE

- 1.1 Name. The name of the nonprofit corporation is Daybreak Eastlake Village Condominiums Owners' Association (the "Association").
- 1.2 <u>Offices</u>. The principal office of the Association shall be located at 5295 South 300 West, Suite 475, Murray, Utah 84107.

ARTICLE II DEFINITIONS

2.1 <u>Definitions</u>. Except as otherwise provided herein or as may be required by the context, all terms defined in Article I of the Declaration of Condominium of Eastlake Village Condominiums (the "Declaration") shall have such defined meanings when used in these Bylaws.

ARTICLE III MEMBERS

- 3.1 Annual Meetings. The annual meeting of Members ("Member" shall have the meaning ascribed to "member of the Association" in Section 5.2 of the Declaration) shall be held within ninety (90) days before or after the close of the Association's fiscal year, beginning with the year following the year in which the Articles are filed, for the purpose of electing Committee Members and transacting such other business as may come before the meeting. If the election of Committee Members shall not be held on the day designated herein for the annual meeting of the Members, or at any adjournment thereof, the Management Committee shall cause the election to be held at a special meeting of the Members to be convened as soon thereafter as may be convenient. The Management Committee may from time to time by resolution change the date and time for the annual meeting of the Members.
- 3.2 <u>Special Meetings</u>. Special meetings of the Members may be called by the Management Committee, the President, or upon the written request of Members holding not less than thirty-three percent (33%) of the Allocated Interests of the Association, such written request to state the purpose or purposes of the meeting and to be delivered to the Management Committee or the President.

LA3:1130245.2

- 3.3 <u>Place of Meetings</u>. The Management Committee may designate any place in Salt Lake County, Utah as the place of meeting for any annual meeting or for any special meeting called by the Management Committee. A waiver of notice signed by all of the members of the Management Committee may designate any place, within the State of Utah, as the place for holding such meeting. If no designation is made, or if a special meeting is otherwise called, the place of the meeting shall be at the principal office of the Association.
- 3.4 <u>Notice of Meetings</u>. The Management Committee shall cause written or printed notice of the time and place, and in the case of a special meeting, the purpose or purposes for all meetings of the Members (whether annual or special) to be delivered, not more than sixty (60) nor less than ten (10) days prior to the meeting, to each Member of record entitled to vote at such meeting. If mailed, such notice shall be deemed to be delivered when deposited in the U.S. mail addressed to the Member at the Member's registered address, with first-class postage thereon prepaid. Each Member shall register with the Association such Member's current mailing address for purposes of notice hereunder. Such registered address may be changed from time to time by notice in writing to the Association. If no address is registered with the Association, a Member's Unit address shall be deemed to be the Member's registered address for purposes of notice in this Section 3.4.
- Owner shall promptly furnish to the Association a certified copy of the recorded instrument by which ownership of such Unit has been vested in such Owner, which copy shall be maintained in the records of the Association. For the purpose of determining Members entitled to notice of or to vote at any meeting of the Members, or any adjournment thereof, the Management Committee may designate a record date, which shall not be more than sixty (60) nor less than ten (10) days prior to the meeting, for determining Members entitled to notice of or to vote at any meeting of the Members. If no record date is designated, the date on which notice of the meeting is mailed shall be deemed to be the record date for determining Members entitled to notice of or to vote at the meeting. The persons or entities appearing in the records of the Association on such record date as the Owners of record of Units in the Project shall be deemed to be the Members of record entitled to notice of and to vote at the meeting of the Members.
- 3.6 Quorum. At any meeting of the Members, the presence of Members holding, or holders of proxies entitled to cast, more than one-half (½) of the Allocated Interests of the Association shall constitute a quorum for the transaction of business; provided, however, that if a quorum is not achieved at the duly called meeting, the Members, including those that are present and represented by proxy, may reconvene the meeting within twenty-four (24) hours if more than thirty-three and one-third percent (33.3%) of the Allocated Interests of the Association are present, in person or by proxy, and transact all business as if there is a quorum.
- 3.7 <u>Proxies</u>. At each meeting of the Members, each Member entitled to vote shall be entitled to vote in person or by proxy; provided, however, that the right to vote by proxy shall exist only where the instrument authorizing such proxy to act shall have been executed by the Member or by the Member's attorney when duly authorized in writing. If a Unit is owned by more than one Person, the instrument authorizing a proxy to act must have been executed by all such Persons or their attorneys when duly authorized in writing. Such instrument authorizing a proxy to act shall set forth the specific matters or issues upon which the proxy is authorized to

- act. Such instrument shall be delivered at the beginning of the meeting to the Secretary of the Association or to such other officer or person who may be acting as secretary of the meeting. The secretary of the meeting shall enter a record of all such proxies in the minutes of the meeting.
- 3.8 <u>Votes</u>. With respect to each matter submitted to a vote of the Members, each Member entitled to vote at the meeting shall have the right to cast, in person or by proxy, the number of votes appertaining to the Unit of such member, as shown in the Declaration. The affirmative vote of a majority of the votes entitled to be cast by the Members present or represented by proxy at a meeting at which a quorum was initially present shall be necessary for the adoption of any matter voted on by the Members, unless a greater proportion is required by the Articles, these Bylaws, the Declaration, the Master Residential Declaration or the Act. The election of Committee Members shall be by secret ballot. When more than one Person owns an interest in a Unit, such Persons shall designate to the Association, in writing, a representative who shall exercise the vote for such Unit on behalf of all co-Owners of the Unit. In no event shall fractional votes be exercised in respect to any Unit.
- 3.9 <u>Waiver of Irregularities</u>. All inaccuracies and irregularities in calls or notices of meetings and in the manner of voting, form of proxies and the method of ascertaining Members present shall be deemed waived if no objection is made either at the meeting or within thirty (30) days of the date of the meeting.
- 3.10 <u>Informal Action by Members</u>. Any action that is required or permitted to be taken at a meeting of the Members may be taken without a meeting, if a consent in writing, setting forth the action so taken, shall be signed by all of the Members entitled to vote with respect to the subject matter of such meeting.

ARTICLE IV MANAGEMENT COMMITTEE

- 4.1 General Powers. The property, affairs and business of the Association shall be managed by a Board of Directors to be known as the Management Committee. The Management Committee may exercise all of the powers and shall comply with all obligations and responsibilities of the Association, whether derived from the Act or the Articles, except such powers that the Articles, these Bylaws, the Declaration, the Master Residential Declaration, or the Act vest solely in the Members. The Management Committee may by written contract delegate, in whole or in part, to a professional management organization or person such of its duties, responsibilities, functions and powers as are properly delegable. Any management agreement must be terminable for cause upon thirty (30) days' notice, have a term not to exceed two years, and may be renewed with consent of the manager and the Management Committee, and if such agreement is entered into during the Declarant Control Period, it must be terminable with or without cause by the Management Committee after the Declarant Control Period ends.
- 4.2 <u>Number, Tenure and Qualifications</u>. The property, business and affairs of the Association shall be governed and managed by a Management Committee composed of not less than three (3) nor more than five (5) persons, each of whom, except for those appointed and serving as Committee Members during the Declarant Control Period must either be an Owner of

3

a Unit, a spouse of an Owner or an agent of Declarant, for so long as Declarant owns a Unit in the Project. At each annual meeting, beginning with the first annual meeting following termination of the Declarant Control Period, the Members shall elect for terms of one (1) year each the appropriate number of Committee Members to fill vacancies by expiring terms of Committee Members. All Committee Members, except such members appointed by Declarant, shall be Members of the Association.

- 4.3 <u>Declarant Control Period</u>. There is hereby set forth a period of Declarant control over the Association (the "**Declarant Control Period**") during which period Declarant or persons designated by Declarant shall have the authority to appoint and remove the Association's officers and Committee Members. The Declarant Control Period shall terminate no later than the earlier of:
 - (a) Six (6) years after the first unit is conveyed to an Owner;
- (b) The later to occur of (i) after the date on which seventy-five percent (75%) of the Units have been conveyed to Unit purchaser, or (ii) after the date on which all the Additional Land has been added to the Condominium Project; or
- (c) When, in its discretion, the Declarant so determines and declares in a written instrument.

At the first annual meeting following expiration of the Declarant Control Period, the Committee Members shall be elected in accordance with these Bylaws.

Notwithstanding anything in these Bylaws or the Declaration that may be construed to the contrary with respect to Declarant's ability to voluntarily terminate the Declarant Control Period as set forth above, Declarant may (in the exercise of its sole and absolute discretion) voluntarily terminate the Declarant Control Period in whole or in part, with respect to all or any portion of any Common Area, any Limited Common Area, any portion of Additional Land, or with respect to any issue, matter or subject whatsoever. Declarant's decision to voluntarily terminate the Declarant Control Period with respect to all or any portion of any Common, Area, Limited Common Area, portion of Additional Land, or with respect to any issue, matter or subject shall in no event affect, modify, or act to waive its authority under the Declarant Control Period except with respect to such Common Area, Limited Common Area, portion of Additional Land, or such issue, matter or subject.

- 4.4 Regular Meetings. The Management Committee shall hold regular meetings at least quarterly, at the discretion of the Management Committee. The Management Committee may designate any place in Salt Lake County, Utah as the place of meeting for any regular meeting called by the Management Committee. A waiver of notice signed by all of the members of the Management Committee may designate any place, within the State of Utah, as the place for holding such meeting. If no designation is made the place of the meeting shall be at the principal office of the Association.
- 4.5 <u>Special Meetings</u>. Special meetings of the Management Committee may be called by or at the request of any Committee Member. The person or persons authorized to call special meetings of the Management Committee may fix any place, within the State of Utah, as the place

4

LA3:1130245.2

for holding any special meeting of the Management Committee called by such person or persons. Notice of any special meeting shall be given at least five (5) days prior thereto by written notice delivered personally, or mailed to each Committee Member at such Committee Member's registered address. If mailed, such notice shall be deemed to be delivered when deposited in the U.S. mail so addressed, with first-class postage thereon prepaid. Any Committee Member may waive notice of a meeting.

- 4.6 Quorum and Manner of Acting. A majority of the then authorized number of Committee Members shall constitute a quorum for the transaction of business at any meeting of the Management Committee. The act of a majority of the Committee Members present at any meeting at which a quorum is present shall be the act of the Management Committee. The Committee Members shall act only as a Management Committee, and individual members shall have no powers as such. A meeting at which a quorum is initially present may continue to transact business, notwithstanding the departure of Committee Members, if at least a majority of the required quorum for that meeting approves any action taken.
- 4.7 <u>Compensation</u>. No Committee Member shall receive compensation for any services that such member may render to the Association as a Committee Member; provided, however, that a Committee Member may be reimbursed for expenses incurred in performance of such duties as a Committee Member to the extent such expenses are approved by the Management Committee.
- 4.8 Resignation and Removal. A member of the Management Committee may resign at any time by delivering a written resignation to either the President or the Management Committee. Unless otherwise specified therein, such resignation shall take effect upon delivery. Any Committee Member, except any such member appointed by Declarant, may be removed at any time, with or without cause, by the affirmative vote of at least fifty-one percent (51%) of the Allocated Interests of the Association at a special meeting of the Members duly called for such purpose. A Committee Member appointed by Declarant may be removed at any time, with or without cause, by the Declarant.
- 4.9 Vacancies and Newly Created Committee Memberships. If vacancies shall occur in the Management Committee by reason of the death, resignation or disqualification of a Committee Member (other than such member appointed by Declarant), the Committee Members then in office shall continue to act, and such vacancies shall be filled by a vote of the Committee Members then in office, though less than a quorum, in any way approved by such Committee Members at the meeting. Any vacancy in the Management Committee occurring by reason of removal of a Committee Member by the Members or if the authorized number of Committee Members shall be increased, such vacancies or newly created Committee Memberships may be filled by election by the Members at the meeting at which such Committee Member is removed or new Committee Membership is created. If vacancies shall occur in the Management Committee by reason of death, resignation or removal of a Committee Member appointed by the Declarant, such vacancies shall be filled by appointments to be made by the Declarant. Any Committee Member elected or appointed hereunder to fill a vacancy shall serve for the unexpired term of his predecessor or for the term of the newly created Committee Membership, as the case may be.

4.10 <u>Informal Action by Committee Members</u>. Any action that is required or permitted to be taken at a meeting of the Management Committee, may be taken without a meeting if a consent in writing, setting forth the action so taken, shall be signed by all of the Committee Members.

ARTICLE V OFFICERS

- 5.1 <u>Officers</u>. The officers of the Association shall be a President, Secretary, Treasurer and such other officers as may from time to time be appointed by the Management Committee.
- 5.2 Election, Tenure and Qualifications. The officers of the Association shall be chosen by the Management Committee annually at a regular meeting of the Management Committee. In the event of failure to choose officers at such regular meeting of the Management Committee, officers may be chosen at any regular or special meeting of the Management Committee. Each such officer (whether chosen at a regular meeting of the Management Committee or otherwise) shall hold such office until the next ensuing regular meeting of the Management Committee and until a successor has been chosen and qualified, or until such officer's death, or until resignation, disqualification or removal in the manner provided in these Bylaws, whichever first occurs. Any person may hold any two or more of such offices, except that the President may not also be the Secretary. No person holding two or more offices shall act in or execute any instrument in the capacity of more than one office. The President, Secretary and Treasurer shall be and remain Committee Members of the Association during the entire term of their respective offices. No other officer need be a Committee Member.
- 5.3 <u>Subordinate Officers</u>. The Management Committee may from time to time appoint such other officers or agents as it may deem advisable, each of whom shall have such title, hold office for such period, have such authority, and perform such duties as the Management Committee may from time to time determine. The Management Committee may from time to time delegate to any officer or agent the power to appoint any such subordinate officers or agents and to prescribe their respective titles, terms of office, authorities and duties. Subordinate officers need not be Committee Members.
- 5.4 <u>Resignation and Removal</u>. Any officer may resign at any time by delivering a written resignation to the President or the Management Committee. Unless otherwise specified therein, such resignation shall take effect upon delivery. Any officer may be removed by the Management Committee at any time, with or without cause.
- 5.5 <u>Vacancies and Newly Created Offices</u>. If any vacancy shall occur in any office by reason of death, resignation, removal, disqualification or any other cause, of if a new office shall be created, such vacancies or newly created offices may be filled by the Management Committee at any regular or special meeting.
- 5.6 <u>The President</u>. The President shall preside at meetings of the Management Committee and at meetings of the Members. The President shall sign on behalf of the

6

Association all conveyances, mortgages, documents and contracts, and shall do and perform all other acts and things as required by the Management Committee.

- 5.7 The Secretary. The Secretary shall keep the minutes of the Association and shall maintain such books and records as these Bylaws, the Declaration or any resolution the Management Committee may require such person to keep. The Secretary shall also act in the place and stead of the President in the event of the President's absence or inability or refusal to act. The Secretary shall be the custodian of the seal of the Association, if any, and shall affix such seal, if any, to all papers and instruments requiring the same, and shall perform such other duties as required by the Management Committee.
- 5.8 The Treasurer. The Treasurer shall have the custody and control of the funds of the Association, subject to the action of the Management Committee, and, when requested by the President, shall report the state of the finances of the Association at each meeting of the Members and at any meeting of the Management Committee. The Treasurer shall perform such other duties as required by the Management Committee.
- 5.9 <u>Compensation</u>. No officer shall receive compensation for any services rendered to the Association as an officer; provided, however, that an officer may be reimbursed for expenses incurred in performance of such duties as an officer to the extent such expenses are approved by the Management Committee.

ARTICLE VI COMMITTEES

- 6.1 <u>Designation of Committees</u>. The Management Committee may from time to time by resolution designate such committees as it may deem appropriate in carrying out its duties, responsibilities, functions and powers. The membership of each such committee designated hereunder shall include at least two (2) Committee Members. No committee member shall receive compensation for services rendered to the Association as a committee member; provided, however, that a committee member may be reimbursed for expenses incurred in performance of such duties as a committee member to the extent that such expenses are approved by the Management Committee.
- 6.2 <u>Proceeding of Committees</u>. The Management Committee may designate one or more committees, each of which shall consist of two or more Committee Members. Each committee designated hereunder by the Management Committee may appoint its own presiding and recording officers and may meet at such places and times and upon such notice as such committee may from time to time determine. Each such committee shall keep a record of its proceedings and shall regularly report such proceedings to the Management Committee.
- 6.3 Quorum and Manner of Acting. At each meeting of any committee designated hereunder by the Management Committee, the presence of members constituting at least a majority of the authorized membership of such committee (but in no event less than two members) shall constitute a quorum for the transaction of business, and the act of a majority of the members present at any meeting at which a quorum is present shall be the act of such committee. The members of any committee designated by the Management Committee

7

LA3:1130245.2

hereunder shall act only as a committee, and the individual members thereof shall have no powers as such. A committee may exercise the authority of the Management Committee subject to Section 16-6a-817 of the Act, as amended, or a similar provision then in effect.

- 6.4 <u>Resignation and Removal</u>. Any member of any committee designated hereunder by the Management Committee may resign at any time by delivering a written resignation to the President, the Management Committee, or the presiding officer of the such committee. Unless otherwise specified therein, such resignation shall take effect upon delivery. The Management Committee may at any time, with or without cause, remove any member of any committee designated by it thereunder.
- 6.5 <u>Vacancies</u>. If any vacancy shall occur in any committee designated by the Management Committee due to disqualification, death, resignation, removal, or otherwise, the remaining members shall, until the filling of such vacancy, constitute the then total authorized membership of the committee and, provided that two or more members are remaining, may continue to act. Such vacancy may be filled at any meeting of the Management Committee.

ARTICLE VII INDEMNIFICATION

- 7.1 <u>Indemnification</u>. No (i) Committee Member, officer, other committee member of the Association, (ii) director, officer, or employee of the Declarant, Master Residential Association or Daybreak Community Council (collectively, the "Indemnitees"), or (iii) employee of the Association (collectively, "Indemnitee") shall be personally liable for any obligations of the Association or for any duties or obligations arising out of any acts or conduct of such Indemnitee performed for or on behalf of the Association. The Association shall and does hereby indemnify and hold harmless each person who shall serve at any time in the same capacity as an Indemnitee, as well as such person's heirs and administrators, from and against any and all claims, judgments and liabilities to which such persons shall become subject, by reason of such Person's capacity relating to the Association, or by reason of any action alleged to have been heretofore or hereafter taken or omitted to have been taken by such person as such Committee Member or officer, and shall reimburse any such person for all legal and other expenses reasonably incurred in connection with any such claim or liability; provided that the Association shall have the power to defend such person from all suits or claims; provided further, however, that no such person shall be indemnified against or be reimbursed for or be defended against any expense or liability incurred in connection with any claim or action arising out of such person's intentional misconduct. The rights accruing to any person under the foregoing provisions of this Section shall not exclude any other right to which such person may lawfully be entitled, nor shall anything herein contained restrict the right of the Association to indemnify or reimburse such person in any proper case, even though not specifically provided for herein or otherwise permitted. The Association, its Committee Members, officers, employees and agents shall be fully protected in taking any action or making any payment or in refusing so to do in reliance upon the advice of counsel.
- 7.2 Other Indemnification. The indemnification herein provided shall not be deemed exclusive of any other right to indemnification to which any person seeking indemnification may be under any Bylaw, agreement, vote of disinterested Committee Members or otherwise, both as

to action taken in any official capacity and as to action taken in any other capacity while holding such office. It is the intent hereof that all Committee Members and officers be and hereby are indemnified to the fullest extent permitted by the laws of the State of Utah and these Bylaws. The indemnification herein provided shall continue as to any person who has ceased to be a Committee Member, officer or employee and shall inure to the benefit of the heirs, executors and administrators of any such person.

- 7.3 Insurance. The Management Committee, in its discretion, may direct that the Association purchase and maintain insurance on behalf of any person who is or was a Committee Member, officer or employee of the Association or is or was serving at the request of the Association as a Committee Member, officer, employee or agent of another association, corporation, partnership, joint venture, trust or other enterprise against any liability asserted against, and incurred by, such person in any such capacity or arising out of such person's status as such, whether or not the Association would have the power to indemnify such person against liability under the provisions of this Article VII.
- 7.4 <u>Settlement by Association</u>. The right of any person to be indemnified shall be subject always to the right of the Association by the Management Committee, in lieu of such indemnity, to settle any such claim, action, suit or proceeding at the expense of the Association by the payment of the amount of such settlement and the costs and expenses incurred in connection therewith.

ARTICLE VIII FISCAL YEAR AND SEAL

- 8.1 <u>Fiscal Year</u>. The fiscal year of the Association shall begin on the first day of January each year and end on the 31st day of December, except that the first fiscal year shall begin on the date of incorporation.
- 8.2 <u>Seal</u>. The Management Committee may by resolution provide a corporate seal which shall be circular in form and shall have inscribed thereon the name of the Association, the state of incorporation, and the words "Corporate Seal."

ARTICLE IX RULES AND REGULATIONS

9.1 Rules and Regulations. The Management Committee may from time to time adopt, amend, repeal and enforce reasonable rules and regulations governing the use and operation of the Project, to the extent that such rules and regulations are not inconsistent with the rights and duties set forth in the Articles, the Declaration, these Bylaws or by law. The Members shall be provided with copies of all rules and regulations adopted by the Management Committee, and with copies of all amendments and revisions thereof.

ARTICLE X AMENDMENTS

10.1 <u>Amendments</u>. Except as otherwise provided by law, by the Articles, by the Declaration, or by these Bylaws, these Bylaws may be amended, altered or repealed and new

Bylaws may be made and adopted by the Members upon the affirmative vote of a majority of the Allocated Interests of the Association; provided, however, that such action shall not be effective unless and until a written instrument setting forth (i) the amended, altered, repealed or new bylaw, (ii) the number of votes cast in favor of such action, and (iii) the total votes of the Association, shall have been executed and verified by the current President of the Association and recorded in the office of the County Recorder of Salt Lake County, Utah. Notwithstanding the foregoing, after the Department of Housing and Urban Development has approved the Condominium Project, these Bylaws may not be amended or the Condominium Project and the Association merged into a successor condominium regime without the prior written approval of the Department of Housing and Urban Development.

Committee Members of Daybreak	EREOF, the undersigned, constituting all of the initial Eastlake Village Condominiums Owner's Association, have day of , 2007.
	<u> </u>
	M. BRUCE SNYDER, Committee Member
	RUSSELL SANFORD, Committee Member
	TY MCCUTCHEON, Committee Member

DECLARANT AND OWNER'S CONSENT

as the owner of the land upon which the C	, 2007, the undersigned, aware corporation, as the Declarant and EVELOPMENT COMPANY, a Delaware corporation ondominium Project is located, do hereby consent to rith the provisions of the Utah Condominium
	KENNECOTT LAND COMPANY, a Delaware corporation
	By: M. BRUCE SNYDER, Vice President, Residential and Commercial Development
	KENNECOTT LAND RESIDENTIAL DEVELOPMENT COMPANY, a Delaware corporation
	By: M. BRUCE SNYDER, Vice President, Residential and Commercial Development

ACKNOWLEDGMENTS

STATE OF UTAH)	
: ss.)	
SNYDER, RUSSELL SANFORD, and TY MC	AKE VILLAGE CONDOMINIUMS OWNERS'
	NOTARY PUBLIC Residing at:
My Commission Expires:	•
STATE OF UTAH) : ss. COUNTY OF)	
M. BRUCE SNYDER, who being by me duly s Residential and Commercial Development of K KENNECOTT LAND RESIDENTIAL DEVEL	ENNECOTT LAND COMPANY and LOPMENT COMPANY, a Utah limited liability is signed on behalf of said companies by express
	NOTARY PUBLIC Residing at:
My Commission Expires:	
	
LA3:1130245.2	1

EXHIBIT "D"

EASTLAKE VILLAGE CONDOMINIUMS

Schedule of Units, Square Footage, Votes and Undivided Interests in Common Areas

Unit Identifying Number	Unit Type	Approx. Sq. Footage of Unit	Undivided Interest Per Unit	No. of Votes Per Unit
Building 1 Unit A	Triplex A	2660	1/30	1
Building 1 Unit B	Triplex A	2369	1/30	1
Building 1 Unit C	Triplex A	1366	1/30	1
Building 2 Unit A	Triplex A	2660	1/30	1
Building 2 Unit B	Triplex A	2369	1/30	1
Building 2 Unit C	Triplex A	1366	1/30	1
Building 3 Unit A	Triplex A	2660	1/30	1
Building 3 Unit B	Triplex A	2369	1/30 -	1
Building 3 Unit C	Triplex A	1366	1/30	1
Building 4 Unit A	Triplex A	2660	1/30	1
Building 4 Unit B	Triplex A	2369	1/30	1
Building 4 Unit C	Triplex A	1366	1/30	1
Building 5 Unit A	Triplex A	2660	1/30	1
Building 5 Unit B	Triplex A	2369	1/30	1
Building 5 Unit C	Triplex A	1366	1/30	1
Building 6 Unit A	Triplex A	1945	1/30	1
Building 6 Unit B	Triplex A	1677	1/30	1
Building 6 Unit C	Triplex A	1366	1/30	1
Building 7 Unit A	Triplex A	1945	1/30	1
Building 7 Unit B	Triplex A	1677	1/30	1
Building 7 Unit C	Triplex A	1366	1/30	1
Building 8 Unit A	Triplex A	1945	1/30	1
Building 8 Unit B	Triplex A	1677	1/30	1
Building 8 Unit C	Triplex A	1366	1/30	1
Building 9 Unit A	Triplex A	1945	1/30	1
Building 9 Unit B	Triplex A	1677	1/30	1
Building 9 Unit C	Triplex A	1366	1/30	1

Building 10 Unit A	Triplex A	1945	1/30	1
Building 10 Unit B	Triplex A	1677	1/30	1
Building 10 Unit C	Triplex A	1366	1/30	1

<sup>All references to building numbers and unit designations refer to such numbers and designations as contained in the Plat.
All square footages are approximate and may vary by up to as much as 10% per unit.</sup>